Automatic defect identification during RRAM characterization

Mathijs Heidekamp







Automatic defect

identification during RRAM characterization

by

Mathijs Heidekamp

to obtain the degree of Master of Science at the Delft University of Technology, to be defended publicly on Wednesday November 15, 2023 at 14:30.

Student number:4485866Thesis committee:Prof. Dr. Ir. S. Hamdioui,
Dr. C. Gao,TU Delft, supervisor
TU DelftDr. Ir. M. TaouilTU Delft
Dr. Ir. M. Fieback,TU Delft, supervisor

An electronic version of this thesis is available at http://repository.tudelft.nl/.



Abstract

Resistive random access memory (RRAM) is an emerging memory technology that has the potential to replace dynamic random access memory (DRAM) or FLASH. The current memory technology suffer from scalability issues. SRAM consumes a lot of area as it requires 6 transistors for a single cell. DRAM stores data using a capacitor which needs to be refreshed periodically. The scalability of DRAM becomes more challenging as the sensing margin becomes smaller and the capacitor needs to be refreshed more frequently increasing the dynamic power consumption. To overcome these challenges new structures, materials and or processes have to be developed which will increase the complexity and cost of DRAM. Flash is volatile memory but has a high operating voltage and relative low endurance. RRAM can be used as potential replacement for Flash and DRAM. RRAM stores information using resistance states instead of charge. RRAM is non-volatile memory, power efficient, scalable and compatible with the current CMOS process.

Before RRAM can be commercialized the quality of RRAM devices needs to be guaranteed. For this we need to be able to diagnose RRAM devices. The diagnosis will allow us to improve the manufacturing process as well as built defect models for memory testing. Traditional fault models use linear resistors to model the manufacturing defects. These fault models do not incorporate the non-linear behaviour of the RRAM device. As a result, tests are created for the wrong test space resulting in a lower yield and more test escapes. To increase the yield and create more reliable fault models, defect models that incorporate the physical defect of the RRAM device are required. To ensure the manufacturing quality of the RRAM devices and for memory testing of RRAM the characterization and diagnosis of RRAM is required. The electrical characterization of RRAM is cheap, fast and is used to evaluate the performance of RRAM. However, none has used electrical characterization for diagnosis to identify defective RRAM devices.

The contribution of this work is to provide the electrical characterization of RRAM to automatically identify defective RRAM devices which can be used for diagnosis. For the first time this is attempted using the electrical characterization of RRAM. For the defect identification, the key electrical parameters are determined. To assess the performance of the defect identification algorithms a method to manually label RRAM devices is proposed. In total five methods have been implemented and evaluated to automatically label defective RRAM devices. A statistical analysis is performed on the key parameters of RRAM. Using the labelled data the devices are compared to a nominal device to identify defects in RRAM. Furthermore, one unsupervised and two supervised learning algorithms have been implemented. K-means, K-NN and a CNN is applied to RRAM devices. The classification algorithms allow the automatic identification of defective RRAM devices which can be used for diagnosis and will contribute to designing defect models that incorporate the non-linear behaviour of the RRAM device. The metrics used for the statistical analysis are insufficient to accurately identify defective RRAM devices. The nominal device method classifies 81% of the device correctly using the euclidean algorithm. The best performance is obtained for the supervised learning algorithm. K-NN classifies 94% of the cycles correct and 84% of the devices. If the data is not labelled the unsupervised learning algorithm can be used. Kmeans classifies 79.6% of the devices correctly which is slightly worse than K-NN. The CNN classifies 67% of the devices correctly using 20 epochs. However, the CNN has not yet been optimized and needs to be improved. For unlabelled data, the unsupervised learning algorithm should be used and for labelled data the K-NN. The electrical characterization of RRAM devices using machine learning looks promising and is much cheaper compared to optical characterization and memory tests. If in the future the defective devices can be linked to the underlying defect, this will lead to cheaper diagnosis of RRAM devices and allow the creation of accurate fault models.

Preface

After a lot of hard work I have been able to complete this thesis. It has not been easy as a lot of implementations and suggestions did not work at first. The goal was to determine which RRAM devices are defective and why but only the first part of this question has been answered. The manual labelling of the RRAM devices required a lot of additional background knowledge and consumed a lot of time. The assumption was that the RRAM characterization using statistics would allow us to identify defects in RRAM devices. However, it turned out that this did not seem possible and we had to switch to machine learning. The additional reading and background information took additional time which is why we have not been able to answer why the devices are defective.

I would like to thank Moritz Fieback for all the support and guidance. I worked a lot from home and the weekly meetings kept me on track and helped me to stay motivated when my suggestions did not work. Additionally, I want to thank my thesis supervisor Said Hamdioui, who provided critical feedback and support during the DfX meetings. I learned a lot from those meetings and would like to thank the members for their suggestions and critical questions during those meetings. I also want to thank Hanzhi Xun for all his help and Hassen Aziza for sharing the measurement data. I would like thank the department for sharing their vocal talents with me during the karoake nights.

I would like to conclude by thanking my mother, brother and girlfriend. Throughout my entire study, but especially during this thesis they have supported me unconditionally!

Mathijs Heidekamp Delft, November 2023

Contents

1	Introduction 1.1 Motivation	1 2 3 4
2	Background 2.1 Overview 2.2 Memory background 2.2.1 Overview 2.2.2 Memory hierarchy 2.2.3 Types of memory 2.2.3 RRAM Background 1 2.3.1 Types of RRAM 1 2.3.2 Switching behaviour of RRAM 1 2.3.3 Technological and electrical parameters 2.3.4 Manufacturing of RRAM devices 1 2.3.5 Defects in RRAM devices	55557001235
3	State of the art in characterization and diagnosis13.1 Overview13.2 Electrical characterization13.2.1 I-V sweeps13.2.2 Pulsed voltage sweeps13.3 Memory tests23.4 Optical characterization23.4.1 TEM23.4.2 AFM23.4.3 SEM23.5 Conclusion2	7 7778011222
4	RRAM data analytics 2 4.1 Overview 2 4.2 Device structure 2 4.3 Metric extraction technique 2 4.3.1 Vset 2 4.3.2 Reset voltage 2 4.3.3 Resistance 2 4.3.4 Switching time set 3 4.3.5 Switching time reset 3 4.3.6 Energy consumption 3 4.3.7 Endurance 3 4.3.8 Variability 3 4.4 Nominal device 3 4.5 Labelling of the data 3 4.6.1 Overview 3 4.6.2 V _{set} 3 4.6.3 V _{reset} 3 4.6.4 LRS 3	5555579111113555779

	4.7	Conclusion	40
5	Defe 5.1 5.2 5.3	ect identification methodologies Overview	41 41 42 43 43 43 44 44
6	Res	ults	45
	6.16.26.36.4	Statistical analysis for defect identification 6.1.1 Results: Statistical analysis 6.1.2 Results: Model I-V curve Unsupervised learning 6.2.1 Pre-processing: Principal component analysis 6.2.2 K-Means Supervised learning 6.3.1 K-Nearest Neighbours 6.3.2 Convolutional neural network Summary	45 45 47 50 50 51 56 56 57 58
7	Disc 7.1 7.2 7.3 7.4	cussion RRAM Data analytics. Classification algorithm. Value of this work. Future work.	59 59 59 59 60
8	Con	clusion	61

Introduction

1.1. Motivation

Reducing the dimensions of current memory technologies below the 10 nm threshold is becoming more difficult due to increased leakage power and structural challenges [1, 2, 3, 4, 5]. Dynamic Random-Access Memory (DRAM) is volatile memory and requires a steady power supply to store information [1, 2, 3]. DRAM stores data using a capacitor that needs to be refreshed periodically. As the dimensions of DRAM are scaled down the capacitance decreases which affects the sensing margin and retention of the device and DRAM cells will need to be refreshed more frequently increasing the dynamic power consumption [1, 3]. To reduce the static and dynamic power consumption new structures, materials and or processes have to be developed which will increase the complexity and cost of DRAM. Additionally, as the dimensions of DRAM are reduced, the leakage power becomes more significant than the dynamic power consumption [1, 2]. Static Random-Access Memory (SRAM) is much faster than DRAM but is also volatile memory and requires 6 transistors to store information making large scale integration costly and infeasible [2, 6]. FLASH is non-volatile memory. However, the endurance is low, operation voltage (10 V) and write speed (1 ms) are high compared to DRAM and SRAM [2, 7]. Additionally, the leakage current is a problem for low power embedded systems. To overcome the scalability challenges and the leakage current alternative materials or memory systems are required [1, 2, 3, 8].

Three promising memory technologies have been developed to overcome these challenges. The emerging memory technologies are phase-change random access memory (PCRAM), spin-transfer torque magnetic random access memory (STT-MRAM) and resistive random access memory (RRAM). The emerging memory technologies are scalable, the read and write speed are better than FLASH memory. Additionally, the emerging memories are non-volatile. However, the focus of this work will be on RRAM. The cell density of STT-MRAM is larger compared to RRAM, the write time of RRAM is faster than of PCRAM and RRAM can be manufactured using the existing CMOS technology. [7].

RRAM is compatible with the CMOS fabrication process and the memory density is high. RRAM is scalable and has zero standby power consumption making it ideal for low power systems [9]. RRAM is not as fast as SRAM so it is unlikely to replace SRAM. However, RRAM can easily be scaled below 1xnm and does not have the scalability issues DRAM is facing [7, 10, 11, 12]. Due to the low leakage current and non-volatile memory RRAM be a good alternative for DRAM. Additionally, RRAM is non-volatile memory and the operation speed of RRAM is much higher than FLASH. The operating voltage is 1 V and operation speed (10 ns) which outperforms the operation voltage and write speed of FLASH [2, 7]. Therefore, RRAM is a good alternative to replace FLASH and DRAM to overcome the leakage power and scalability issues of current memory technologies.

Before RRAM can be commercialized we need high quality tests that can identify defects in RRAM devices. A defect is an unintended variation in the design. Defects are modelled using fault models. A fault model in an abstraction of a defect that describes the faulty behaviour at the electrical level [13]. Current test methods use fault models that mimic physical defects using linear resistors [14, 15, 16, 17]. These models do not properly mimic the non-linear behaviour of RRAM and do not catch the unique faults that occur in RRAM leading to test escapes [14, 18]. The unique faults that are Hard-to-Detect occur occasionally or only result in parametric variations affecting the long-term reliability of the device

[9]. In order to commercialize RRAM devices the unique faults that can occur need to be identified and understood so that accurate fault models can be built and new test methods can be designed. To be able to build accurate fault models the manufacturing defects need to be identified and linked to the behaviour of the device. The manufacturing defects and the effect on the switching characteristics have already been investigated by Fieback, Taouil, and Hamdioui [14]. So far we have only attempted to identify the defects using fault models which are an abstraction of the defect and cannot be linked back to the actual defect. Diagnosis of RRAM devices is required to determine the cause of the defective behaviour.

1.2. State of the art

Defect characterization of RRAM devices is necessary to develop accurate fault models capable of detecting unique defects in RRAM. To identify the process variations and manufacturing defects the electrical parameters of the device need to be determined and the effect of the manufacturing defects on the electrical parameters needs to be studied [14]. Fieback, Taouil, and Hamdioui have already identified the defects that can occur during the manufacturing process and proposed the key parameters that should be used for characterization [14]. Lanza et al. proposed similar parameters together with a method to extract these parameters from measurement data and the figure of merits that should be used for visualization. The electrical parameters describe the switching behaviour of the device [19]. While there is no discussion about the electrical parameters, the method to extract and analyse these parameters is unclear. Lanza et al. does propose the figure of merits for each of the electrical parameter [19]. The figure of merits give information about the standard deviation, average and distribution of the RRAM cells. However, they do not mention how a healthy device behaves and how defects affects these parameters. The effect of the manufacturing defects on the parameters is difficult to assess due to the cycle-to-cycle and device-to-device variability. Poehls et al. do mention the physical defects that can occur during the manufacturing process [9]. But how exactly this affects the parameters or how the defects can be identified is unclear. The random misbehaviour in some cycles makes defect identification and fault diagnosis difficult as there is no baseline that mentions the minimum number of switching cycles that need to be analyzed to assess the performance of the device and whether the behaviour is due to the variability of the device or is caused by a defect. Additionally, the visual inspection of individual cycles is time-consuming and subjective, and there is no guarantee that the parameters can be used to identify defective behaviour. As a result, a reliable method is required that automatically extracts the key features of a RRAM cell and classifies the device.

The characterization of RRAM devices can be split into three categories. Electrical characterization, memory testing and optical characterization. Electrical characterization characterizes RRAM using electrical measurements from RRAM cells. Memory testing attempts to identify defective behaviour in RRAM using fault models and optical characterization characterizes the device using optical measurements.

Electrical characterization is much cheaper than optical characterization. In literature, there are two main methods for the electrical characterization of RRAM. Voltage sweeps (I-V sweeps) where the current of the device is measured while the voltage is gradually increased and pulsed voltage stress (PVS) in which short voltage pulses are applied to the device to make the device switch between states [2, 19]. Zahoor, Zulkifli, and Khanday has used I-V sweeps to study the impact on device dimensions [2]. Fantini et al. analysed the impact of oxide thickness on the switching characteristics by applying PVS [20]. Grossi et al. studied the effect of grain boundaries in polycrystalline and amorphous HfO₂ [21]. While these give information about how the physical properties of RRAM affects the device behaviour they do not discuss how unwanted parametric deviations in the physical design can be identified. The identification of these defects using electrical tests is cost effective, fast and will result in more accurate fault models leading to fewer test escapes.

Memory testing is cheaper than the electrical characterization, but more abstract compared to the electrical tests. Memory tests use fault models. A fault is an abstraction of a defect that describes the faulty behaviour at the electrical level. In literature a distinction between conventional and unique faults is made. Conventional faults are faults that can also be observed in mainstream memories and unique faults are new emerging faults caused by the analog characteristics of RRAM. March tests have been proposed by many. However, none are able to identify the unique faults in RRAM devices as the fault models are inaccurate leading to test escapes. Tests do not accurate catch the faults in RRAM devices

and are not linked to the unique defects [22, 23, 24]. Chen et al. and Liu et al. proposed a march test to identify unique faults in RRAM. In [22], read disturb faults in RRAM are identified, but the link to the physical defect is missing. In [23] they were able to identify faulty cells, but the root-cause of the fault is lacking. Kannan, Karri, and Sinanoglu uses sneak-paths to identify defects in RRAM but the effect of the defects on the parameters is not discussed. The problem is that the fault models do not incorporate the analog characteristics of RRAM. Fieback et al. proposes the device-aware-test which instead of the linear defect model by modelling the defective behaviour of the defect. However, to built a physical defect model we first have to identify which RRAM devices show abnormal behaviour and what defect causes the abnormal behaviour.

Optical characterization is the most expensive. For the optical characterization transmission electron microscopy (TEM), atomic force microscopy (AFM) and scanning electron microscopy (SEM) is used. The optical characterization has been used to study the switching of RRAM in situ as well as the effect of defects[26, 27, 28, 29]. Pey et al. studied the effect of the compliance current on the switching of RRAM cells using in-situ TEM. Nandi et al., Charpin-Nicolle et al. used AFM to measure the electrode roughness of RRAM cells. In [32] CAFM is used in-situ to study the effect of grain boundaries. Optical characterization visualizes the physical defect. However, the dimensions of the device affect the switching behaviour of RRAM. For TEM a thin cross-section is required for the examination of RRAM devices. As a result, the switching behaviour of the device is not comparable to that of typical devices the dimensions of the device are not representable. Furthermore, the number of cycles that can be characterized is limited and due to the high current density the number of cycles that can be characterized is limited and in some cases destroys the device [19].

Visualizing the defect using optical characterization is expensive, time consuming and cannot be used at array-level. The memory tests use faults to identify defective behaviour. As the faults are an abstraction of the actual defect it is good to use for identifying faulty devices, but cannot be used for diagnosis as the physical defect cannot be retrieved. Electrical characterization is cheap as we only require measure data. So far the electrical characterization has only been applied to study the impact of device dimensions and has not yet been used for the characterization and diagnosis RRAM.

1.3. Contribution of the work

This master thesis focuses on the automatic defect characterization of RRAM devices using the key parameters proposed by Lanza et al. and Fieback et al. to identify functional and defective RRAM devices [15, 19]. The electrical parameters describe the switching characteristics of RRAM. The methods to extract the electrical parameters will be evaluated and compared. A statistical analysis will be done on the electrical parameters of 49 devices and 936 cycles to assess the performance of the device. The statistical analysis will give insight in the distribution and behaviour of the different devices, which will be used to classify healthy and defective devices. Besides the statistical analysis, three machine learning techniques are used and applied to the IV measurements and the electrical parameters. To assess the performance of these methods, the devices are manually labelled for comparison. The method to do this is presented as well. This thesis presents a method to identify defects in RRAM devices using statistical and machine learning methods. To summarize the main contributions.

- 1. Identification of the key electrical parameters of RRAM devices
- 2. Comparison between different methods to extract the electrical parameters from RRAM devices
- 3. Extensive statistical analysis of the electrical parameters to characterize RRAM devices.
- 4. Framework to manually label RRAM devices which can be used to evaluate the performance of classification algorithms.
- Implementation of one unsupervised and two supervised machine learning algorithms that can identify defective RRAM devices.
- 6. Software to automatically extracts the electrical parameters and identify defective RRAM devices.
- Contribution to a paper that is to be presented at the Asian Test conference 2023. The software provided in this work aided in the analysis of the data. H. Xun et al. "Characterization and Test of Intermittent Over RESET in RRAMs". In: ATS conference 2023.

1.4. Outline

Chapter 2 provides an comparison of the state-of-the-art memory and emerging memory technologies. Furthermore, background information on RRAM is provided including the steps required to manufacture RRAM devices and the defects that can occur in the process. Chapter 3 presents the state-of-the-art in characterization and diagnosis in the microelectronics field and what is missing. Chapter 4 describes the measurement data, the method to extract the electrical parameters, the method to label RRAM devices and provides an initial characterization of the RRAM devices using statistics. Chapter 5 provides the algorithms that are used to classify the different RRAM devices. Three different methods are described. The statistical method, one unsupervised learning methods and two supervised learning algorithm. Chapter 6 shows the results of the different algorithms. Chapter 8 evaluates the results and provides a summary of the results and evaluates the performance of the algorithms. Chapter 7 evaluates the reliability of the work. What is the contribution to the scientific community, are the results reliable and what should be the next step.

Background

2.1. Overview

This chapter provides the background in memory and RRAM to understand the importance of RRAM characterization and defect identification. The first part explains the memory hierarchy, the fundamentals of the different memory types and provides a comparison between the different memory types in terms of performance and cost. The second part of the chapter focuses on the background information of RRAM. The hardware layout, the electrical and technological parameters and the basic principles of RRAM. The chapter will end with the fabrication process and the corresponding manufacturing defects in RRAM.

2.2. Memory background

2.2.1. Overview

The performance gap describes the increasing difference in processing speed and latency between the CPU and memory system [33]. The performance gap between CPUs and memory is growing at a rate of 50% per year [33]. The performance gains of CPUs and memory over the past three decades are illustrated in Figure 2.1 [33, 34]. Figure 2.1 shows the increasing performance gap between memory and CPU indicating the recent developments of the CPU and memory. From 1990-2000 the lower level caches are introduced to reduce latency. Then multi-cores where developed which allowed the execution of multiple instructions in parallel. Each core was introduced with lower L1 and L2 cache and the L3 cache is introduced that is shared by the multiple cores. As the number of cores grew the L3 cache was distributed over the individual cores. The last development for the CPU to improve the performance are XPUs. An XPU is an additional unit that works together with a CPU. It is dedicated hardware designed for a specific task which can execute that task faster than a CPU [33]. Furthermore for Memory DDIO (Data Direct Input Output) was designed for networking. It allows direct placement of the data in the processor cache without having to place it first in memory. After which the guality of service (QoS) for cache was implemented to reduce resource contention between the different cores. The memory wall is a manifestation of the performance gap and implies the effects of teh growing performance gap between the CPU and memory. To adress the memory wall and minimize latency the memory hierarchy was introduced incorporating different types of memory [33].

2.2.2. Memory hierarchy

The memory hierarchy integrates different types of memory to reduce latency and increase bandwidth [33, 34, 35, 36]. Memory types are organized based on their size, cost, and access speed. The farther the memory is from the processor, the slower it becomes, but in return offers a larger storage capacity. Conversely, memory located close to or directly on the CPU is faster but has limited storage space.

The memory hierarchy shown in Figure 2.2 consists of several levels: registers, cache, main memory, flash, secondary storage and remote storage [37]. The figure shows the growing performance gap between CPUs and memory from 1990 up til 2020 and the attempts to reduce the performance gap. The development of the CPU went from single thread to multi-threading to the use of accelerators in-



Figure 2.1: Yearly improvement of processor and DRAM memory speeds, for a time period of three decades obtained from [34]

creasing the performance gap between memory and the CPU further. To overcome this cache memory is introduced that is placed closer to the CPU. Initially a single thread CPU was used with off-die level 1 cache consisting of a couple hundred of Kilobytes. Over time the lower levels cache moved on chip and can now store more than 100 MB [34]. The components of the memory hierarchy are:

- Registers are located on the CPU and are the fastest memory type. The storage capacity is typically a few of bytes. Registers are built using flip-flops which require continuous power to maintain their state [36]. Due to the limited space on the CPU, expanding the register size is not feasible. Increasing the size would require more interconnects, resulting in a higher power consumption cost and circuit complexity. This would contradict the objective of having a low power, high performance device.
- 2. Cache memory is situated close to or directly on the CPU chip and can store more data than registers. Cache memory serves as a bridge between the CPU and main memory, holding frequently accessed instructions and data. Cache memory is typically constructed using Static Random Access Memory (SRAM) and is volatile memory. The memory size can vary between several kilobytes up to a few of megabytes. Data required by the CPU is copied in chunks from secondary storage to cache, assuming that the data is spatially distributed [36, 38].
- 3. Main Memory is larger than cache memory and is built using the same type of memory. However, it differs in terms of capacity and speed. As a result, main memory is not integrated onto the CPU, with the exception of embedded systems, but is housed in a separate module. Main memory predominantly holds the programs executed by the system, while cache memory stores data frequently used by the CPU. Dynamic Random Access Memory (DRAM) is commonly used for main memory [36, 39].
- 4. Flash Disk is a type of non-volatile memory designed for data storage. Non-volatile memory refers to memory that can retain stored data even when the power is disconnected. Unlike volatile memory, such as SRAM and DRAM, which loses data when the power supply is interrupted, non-volatile memory retains information over long periods. Flash memory, which falls into the non-volatile memory category, offers slower access times compared to cache memory and is placed further away from the CPU. It is commonly utilized in embedded systems such as cameras, smartphones, and USB drives. Flash memory is re-programmable, allowing for data to be written and erased multiple times. This characteristic, combined with its ability to retain data without a power source, makes flash memory a popular choice for applications requiring persistent storage [7, 40].

- 5. Secondary storage refers to the main hard drive used to store data that is not immediately required by the CPU. It can accommodate large amounts of data but is extremely slow [41].
- 6. Remote storage is not part of the local memory hierarchy and can be accessed through a network. This is the slowest type of memory within the memory hierarchy. The advantage is that the data can easily be accessed by multiple users, which makes it ideal for data storage, backup and cloud computing.

Overall, the memory hierarchy optimizes the balance between storage capacity and access speed, placing faster and smaller memory types closer to the CPU and utilizing slower but larger storage further away.



Figure 2.2: The memory hierarchy [37]

2.2.3. Types of memory

subsection 2.2.2 introduced the memory hierarchy. This section provides an overview of the memory types used in the memory hierarchy. SRAM, DRAM, flash, and magnetic storage are discussed to offer a more comprehensive understanding of their characteristics. The hardware layout, advantages, disadvantages, and limitations associated with each memory type are discussed.

SRAM

SRAM is faster compared to DRAM at the expense of area and cost. A typical SRAM cell is given in Figure 2.3a. A typical SRAM cell consists of six transistors. Two are used for the selection of the word and bit line and the other four transistors are cross-coupled inverters that store the state of the device [2, 6]. The additional transistors make large-scale integration costly and infeasible [2, 6]. Table 2.1 shows the device characteristics which shows that the read and write time is 10 times better compared to DRAM but that the area required is much larger with a cell area of $100F^2$ compared to $6F^2$ where *F* is the feature size of the lithography method used. Due to the performance, SRAM is used in cache memory which is placed close to the CPU [2, 7].

DRAM

DRAM is volatile memory. DRAM offers greater area efficiency as it only requires a single transistor and capacitor to store information [1, 2, 3]. Figure 2.3b shows the DRAM circuit. A charged capacitor represents a logical '1' and an uncharged capacitor represents a logical '0'. The capacitor leaks charge over time. As a result, the DRAM cell needs constant refreshing to be able to distinguish a logical '1' from a logical '0'. The capacitance of the capacitor is determined by the plate dimensions and dielectric constant. However, scaling down DRAM cells presents challenges. The reduced size decreases the capacitance and makes the sensing margin smaller [1, 3]. Additionally, the dynamic power consumption increases because the capacitor needs to be refreshed more frequently to maintain its state. To further scale down DRAM devices, the development of new structures, materials, or processes becomes necessary. However, these advancements result in increased complexity and higher costs associated with DRAM production.

Flash

Flash uses floating gate transistors and utilizes the threshold voltage of transistor gates to store information. Figure 2.3c shows a flash memory cell. Flash memory is non-volatile memory that retains data even when the power is turned off. To write data to the cell, a high voltage is applied to the transistor gate, creating an electric field that allows electrons to flow through the oxide into the floating gate. To erase data a voltage is applied to the entire memory cell, causing the electrons to move back from the transistor gate to the source. As shown in Table 2.1, flash memory is considerably slower than DRAM and SRAM. However, its non-volatile characteristic makes it well-suited for long-term storage data in embedded devices and solid-state drives. Another limitation of flash is that the endurance is low compared to SRAM and DRAM and the write voltage is high [7, 40].



Figure 2.3: Hardware layout of SRAM, DRAM and flash [42, 43].

Emerging memory

SRAM has fast read and write speeds but is limited by its low density, high cost and volatile characteristics [2]. Scaling down DRAM below the 1xnm threshold is challenging and expensive, and also suffers from high leakage current as well as dynamic power consumption [45, 46]. Flash is non-volatile memory and cheap to implement but suffers from drawbacks such as high operating voltage, slow write and read speeds, and limited endurance. These limitations drive the need for new memory technologies that offer improved scalability, lower leakage current and reduced costs. The emerging memory technologies that show potential are phase-change random access memory (PCRAM), spin-transfer torque magnetic random access memory (STT-MRAM) and resistive random access memory (RRAM). These memory technologies are discussed in more detail in this section.

PCRAM

PCRAM consists of two electrodes, a dielectric isolator and a phase change material. The phase change material has the ability to switch between two phases: amorphous and crystalline [47, 48]. By applying a large electrical pulse to the device, the material heats up and switches from the crystalline to the amorphous state. By applying a longer medium current pulse the crystalline state is restored [48]. The resistance of the device changes depending on the state of the phase change material. PCRAM offers several advantages. The read time is comparable to DRAM while having non-volatile memory similar to flash. The operation voltage is lower with 3 V compared to the 10 V required for flash and as is shown in Table 2.1 the endurance is much better compared to flash memory.

STT-MRAM

STT-MRAM uses a magnetic tunnelling junction (MTJ) to store data. The MTJ consists of two ferromagnetic layers: the reference layer and free layer which are separated by a dielectric material [49]. The reference layer has a fixed magnetic direction, while the magnetic field of the free ferromagnetic layer can be altered by an external electromagnetic field. The value read or stored from the STT-MRAM cell depends on the relative orientation between the reference layer and the free layer. STT-MRAM is a non-volatile memory that retains data when there is no power. The operation voltage of 1.5 V is 0.5 V higher than SRAM and DRAM with comparable endurance, with an expected lifespan of more than 10^{16} write cycles as can be seen in Table 2.1. Making it a perfect candidate to replace flash.

RRAM

RRAM consists of a bottom, a top electrode, and an insulator in between. When a voltage is applied across the electrodes, a filament is formed within the insulator, connecting two electrodes. By applying a voltage of opposite polarity, the filament is broken down. The conducting filament (CF) allows current to flow, reducing the resistance over the electrodes. The resistance of the RRAM device varies based on the length of the CF to which a logical one or zero can be assigned [2, 7]. Compared to DRAM and flash, RRAM exhibits superior scalability below the 10 nm threshold as can be seen in Table 2.1. RRAM also has better endurance than flash, while occupying a smaller cell area. Moreover, RRAM demonstrates comparable read and write speeds to DRAM. Thanks to its non-volatile nature, RRAM can serve as a viable replacement for flash memory. Furthermore, in low power embedded systems where zero leakage current is desired, RRAM might even be used as a substitute for DRAM. However, new materials and additioanl manufacturing steps are required to fabricate RRAM. These manufacturing steps give rise to new faults for which specialized tests have to be developed.



Figure 2.4: Hardware layout of PCRAM, STT-MRAM and RRAM

Summary

The device characteristics of different memory types are presented in Table 2.1. The emerging memory technologies, including RRAM, STT-MRAM, and PCRAM, demonstrate superior scalability and improved cell area compared to current state-of-the-art memory technologies. While SRAM remains the top performer in terms of performance, RRAM, STT-MRAM, and PCRAM offer potential as replacements for Flash memory and as competitors to DRAM due to their volatile nature. Currently, the emerging memory technologies are not yet used for large scale production. The emerging memory technologies require different manufacturing steps giving rise to new defects. To catch and identify these defects new tests have to be developed before these memory types can become mainstream memory types.

Based on Table 2.1, both STT-MRAM and RRAM are promising replacements. However, the cell density of STT-MRAM is larger compared to RRAM. RRAM devices can be manufactured using existing CMOS technology and the available data we have comes from RRAM cells [7]. Therefore, this master's thesis work will focus on RRAM as a potential replacement for traditional memory types.

	Mainstream memories				Emerging memories			
			Fla	Flash				
	SRAM	DRAM	NOR	NAND	STT-MRAM	PCRAM	RRAM	
Cell area	$> 100F^2$	$6F^{2}$	$> 10F^2$	$< F^{2}(3D)$	$6 - 50F^2$	$4 - 30F^2$	$4 - 12F^2$	
Multibit	1	1	2	3	1	2	2	
Voltage	< 1V	< 1V	> 10V	> 10V	< 1.5V	< 3V	< 3V	
Read time	$\sim 1ns$	$\sim 10 ns$	$\sim 50 ns$	< 10µs	< 10 <i>ns</i>	< 10 <i>ns</i>	< 10 <i>ns</i>	
Write time	$\sim 1ns$	$\sim 10 ns$	$\sim 10 \mu s - 1 m s$	$100\mu s - 1ms$	< 10 <i>ns</i>	$\sim 50 ns$	< 10 <i>ns</i>	
Retention	N/A	$\sim 64ms$	> 10 <i>y</i>	> 10y	> 10 <i>y</i>	> 10 <i>y</i>	> 10 <i>y</i>	
Endurance	> 1e16	> 1e16	> 1e5	> 1e4	> 1e15	> 1e9	> 1e6 – 1e12	
Write energy (J/bit)	$\sim fJ$	$\sim 10 fJ$	$\sim 100 pJ$	$\sim 10 fJ$	$\sim 0.1 pJ$	$\sim 10 pJ$	$\sim 0.1 pJ$	

Notes: F: feature size of the lithography. The energy estimation is on the cell-level (not on the array-level). PCRAM and RRAM can achieve less than $4F^2$ through 3D integration. The numbers of this table are representative (not the best or the worst case)

2.3. RRAM Background

Resistive Random Access Memory (RRAM) has a metal-insulator-metal structure [2, 7].Figure 2.5 illustrates the schematic of a typical RRAM device. The resistance of the device can be altered through electrical stress [19]. By applying a positive voltage to the device, a conducting filament (CF) forms within the insulator, reducing the resistance of the device. In the case of bipolar switching, the polarity is reversed to disrupt the CF. resulting in an increase in resistance. In the case of unipolar switching the resistive state is controlled by the same polarity [2, 19].

2.3.1. Types of RRAM

Conductive bridge RAM (CBRAM) and Metal Oxide Resistive Memory (OxRAM) are two distinct variations of RRAM that differ in the type of insulator material used. OxRAM utilizes an oxide as the insulator, whereas CBRAM employs a solid electrolyte. In OxRAM, the formation of the conducting filament (CF) occurs through the creation of oxygen vacancies within the oxide. CBRAM forms a CF through the migration of metal ions into the electrolyte [7]. One advantage of CBRAM is that the metal atoms can recombine more rapidly with the solid electrolyte, thereby enhancing the switching speed. However, this comes at the expense of retention and endurance [7]. Retention refers to the duration for which data can be reliably preserved, while endurance denotes the number of reliable switching cycles a device can endure. Given that the available measurement data pertains to OxRAM and considering its improved retention and endurance characteristics, our focus will be directed towards OxRAM devices.



Figure 2.5: RRAM layout [2]

2.3.2. Switching behaviour of RRAM

To be able to identify defective behaviour in RRAM the switching process of RRAM needs to be understood. The focus lies on OxRAM, but the principle is similar for CBRAM. Figure 2.6 illustrates the different states of RRAM [51].

Step A: The unformed device after fabrication. The device consists of two electrodes: a top and bottom electrode, separated by a metal-oxide containing a few oxygen vacancies which acts as a semiconductor. Typically there is a capping layer just below the top electrode (TE) where the oxygen atoms are stored. A soft breakdown is required before the device can be used [51].

Step B: During the soft breakdown phase a positive voltage is applied to the device. The voltage causes the metal-oxide bonds to break. The oxide atoms migrate towards the capping layer leaving oxygen vacancies in the metal-oxide. The oxygen vacancies form the conductive filament (CF) that connects the two electrodes, decreasing the resistance of the metal-oxide. After this process the device is in a low resistive state [51].

Step C: In the low resistive state (LRS) the filament connects the top and bottom electrode. Thanks to the CF, the resistance of the device is low. The state is considered as a logical '1'.

Step D: During the reset process a voltage with opposite polarity is applied to the device. The voltage causes the oxygen atoms to migrate from the capping layer to the metal oxide. The oxygen atoms migrate from the capping layer back to the metal-oxide. The metal-oxide bonds are restored and the oxygen vacancies in the metal-oxide fare filled. The CF is broken down, which increases the resistance of the RRAM cell. After this process the device is in a high resistive state [51].

Step E: In the high resistive state (HRS) the filament no longer connects the top and bottom electrodes. There is a gap between the filament and the top electrode. The conductive properties of the metal-oxide is lower than that of the oxygen vacancies of the CF. As a result, the resistance of the RRAM cell is several orders of magnitude higher. This state represents a logical '0' [51].

Step F: During the set process a positive voltage is applied to the device. The oxidized metaloxide bonds break and migrate back to the capping layer. As a result, the gap between the CF and the top electrode disappears, and the CF re-establishes the connection between the two electrodes. The voltage during this process is lower compared to the initial soft breakdown, as the filament has already been partially formed. After the set process the device is in a low resistive state [51].



Figure 2.6: RRAM switching behaviour obtained from [52] who adapted it from [51]

2.3.3. Technological and electrical parameters

The characteristics of RRAM devices are described by technological and electrical parameters. The technological parameters describe the device structure and the physical properties of the RRAM cell. The electrical parameters describes the electrical behaviour of the device. The electrical parameters are affected by the device structure and the electrical properties of the used materials. The electrical and technological parameters are used to characterize the device.

Technological parameters

The technological parameters of an OxRAM device as described by Fieback et al. are presented in Table 2.2 [15]. Figure 2.7 gives a schematic view of the technological parameters. The technological parameters include the thickness of the oxide (t_{ox}) , the length of the conductive filament (l_{CF}) , the gap between the top electrode and the CF (l_{gap}) , the width of the top of the CF (ϕ_T) and the width of the bottom of the CF (ϕ_R) .

Each technological parameter impacts the behaviour of the RRAM cell [9, 14, 19]. The oxide thickness affects the resistance distribution of the device. The presence of grain boundaries in the oxide has a mitigating effect on the resistance distribution [32, 53]. A grain boundary is a boundary between two crystallites (grains). The structure of a single crystallite is highly ordered. This structure can be disrupted at the boundary by an adjacent crystallite when the two crystals are not aligned relative to each-other [54]. The disruption of the atomic structure at the boundary affects the electrical properties of the crystalline structure [32]. At these boundaries the number of oxygen vacancies is larger. As a result the conductive filament tends to form along these grain boundaries affecting the switching characteristics of the RRAM device. As the size of RRAM devices decrease the impact of single grain boundaries increase the device-to-device variability. Consequently, a thicker oxide reduces the device variability. Additionally, a higher forming voltage is required during the soft breakdown process to connect the two electrodes in LRS.

The gap length and the length of the filament are antagonistic parameters [15]. As the filament length increases, the gap length decreases. The gap affects both the switching voltage and the resistance of the device. A larger gap length results in a higher resistance in HRS. However, this will also lead to longer switching times, as more time is required to transition to the LRS.

The thickness of the filament impacts the retention and variability of the RRAM device [15]. In the high resistive state, the residual conductive filament can serve as a source of vacancies that bridge the gap between the filament and the top electrode, thereby reducing the retention of the device. However, a thicker filament leads to decreased variability, affecting both the voltage and resistance distributions of RRAM.

Electrical parameters

The electrical parameters describe the electrical properties of a single RRAM cycle. The parameters are described in Table 2.2.

- *V_{reset}* is the voltage at which the device switches from a low resistive state to the high resistive state.
- V_{set} is the voltage at which the device initiates the transition from a high resistive state to low resistive state.
- *HRS* is the resistance of the device in high resistive state.
- *LRS* is the resistance of the device in low resistive state.
- $t_{H \Rightarrow L}$ describes the time required to switch from high resistive state to low resistive state
- $t_{L \Rightarrow H}$ describes the time required to switch from low resistive state to high resistive state

The electrical parameters are influenced by the corresponding technological parameters. Thicker and larger devices have a higher switching probability, which affects the reset and set voltage. As mentioned previously, the thickness and length of the conductive filament also affect the resistance distributions of LRS and HRS [55]. The electrical parameters are mainly used to study the characteristics and behaviour of the RRAM device [15, 19].

	Technological parameters		Electrical parameters
t_{ox}	Oxide thickness	Vreset	Reset threshold
l_{CF}	CF length	Vset	Set threshold
l_{gap}	Gap length	HRS	Reset resistance
ϕ_T	CF top width	LRS	Set resistance
ϕ_B	CF bottom width	$t_{H\Rightarrow L}$	HRS to LRS delay
		$t_{L \Rightarrow H}$	LRS to HRS delay

Table 2.2: Key technological and electrical RRAM parameters

Figure 2.7: Schematic view of technological parameters RRAM device obtained from [15].



2.3.4. Manufacturing of RRAM devices

Overview

The manufacturing process of RRAM devices consists of two main stages: The front-end-of-line (FEOL) and back-end-of-line (BEOL) [14, 15]. The FEOL describes the construction of the transistor. The manufacturing process in the FEOL stage has been studied extensively and refined, resulting in a streamlined process. The defects that can occur are known and comprehended. The BEOL describes the construction of the interconnects and any passive device which in this case is the RRAM cell. This stage of the fabrication process is less reliable and the impact of defects on the RRAM devices has not yet been fully characterized. To get a better understanding of the unique faults that can occur in RRAM devices, the effects of the defects on the electrical parameters need to be studied in more detail.

Front-end-of-line

The front-end-of-line stage(FEOL) describes the fabrication of transistors, which is similar to the conventional CMOS production. To construct the transistors on a single-crystalline silicon wafer the following processing steps are required: Photolithography, etching, doping ,material deposition and planarization. After the explanation of the different processing steps during FEOL an overview of the corresponding defects will be discussed [56].

- 1. During photolithography a photoresist coating of about 1 µm is created to protect specific areas of the wafer during the etching process. A mask consisting of transparent and opaque regions is placed above the wafer. The wafer and the mask are then exposed to ultra-violet light. Assuming a negative photoresist coating, the UV-light makes the transparent regions insoluble. The wafer is exposed to a base or acid to remove the photoresist that has not been exposed to the UV-light and soft-baked to harden the remaining photoresist [56].
- 2. Etching is performed to remove areas of the wafer that are not protected by the photoresist coating. The etching is done using chemical solutions [56].

- 3. Doping modifies the electrical properties of the exposed material. Atoms are introduced to the material which alters the electrical characteristics of the material. There are two methods: Diffusion and ion implantation. In diffusion implantation, dopants diffuse into the exposed regions of the wafer under high temperatures. The disadvantage is that you have limited control over the distribution as the dopant concentration is highest at the surface and decreases with a guassian profile as you go deeper into the wafer. In ion implantation, the dopants are introduced as ions to the exposed materials. Based on the duration and the acceleration of the ions you can control the dept and concentration of the dopants in the exposed material. Ion implantation is more expensive and an additional annealing step is required to repair the displacement of substrate atoms[56].
- 4. Material deposition creates thin layers of different materials that are stacked on top of each other. This can be done through physical vapor deposition or chemical vapor deposition[56].
- 5. Material deposition creates thin layers of different materials that are stacked on top of each other. This can be done through physical vapor deposition (PVD) or chemical vapor deposition (CVD). Chemical vapor deposition uses a chemical reaction in combination with a gas-phase precursor to deposit a solid film on a substrate. This occurs at high temperatures. Physical vapor deposition uses a physical process either sputtering or evaporation which evaporates the material in a vacuum after which the material evaporates on the thin film. The technique depends on the type of material. For CVD higher temperatures are required and is generally used for depositing dielectric materials. PVD is generally used for the metal layers and interconnects as the PVD techniques provides more accuracy, but is also more expensive[56].
- 6. Planarization is done to flatten and smoothen the surface of the wafer. This step is necessary to address the rough topography caused by the material deposition process[9, 15, 56].

Back-end-of-line

In the back-end-of-line (BEOL) stage, the interconnects for both the transistor and the RRAM device are built on the wafer. Several processing steps are required to build the RRAM device introducing different defects than the traditional CMOS production process [9].

- 1. Metallization involves applying a thin-film metal layer to create interconnects between the different components of the integrated circuit [9].
- 2. Bottom electrode patterning, the bottom electrode is deposited and patterned so that it connects to the underlying metal layers. Generally this is done using physical vapor deposition. PVD leaves a rough electrode surface which needs to be planarized [9].
- 3. BE planarization is required to reduce the surface roughness of the BE created by the PVD during patterning [9].
- 4. Metal-oxide deposition deposits the oxide on top of the BE. The thickness of the oxide affects the electrical parameters of the device so this process needs to be carefully controlled. As a result, atomic layer deposition (ALD) is used to place the oxide on top of the BE. ALD is more accurate than physical vapour deposition, which improves the uniformity and reduces the variability of the device [9].
- 5. Cap deposition is required to store the free oxygen atoms from the metal-oxide [9].
- 6. TE deposition is similar to the bottom electrode patterning. The top electrode is placed on top of the capping layer using physical vapour deposition [9].
- 7. Post-processing methods are applied to finish the RRAM device. Passivation is performed to protect the semiconductor from interaction with air or other materials. Etching is done to remove the parts that are not masked. An additional metallization layer is added to finalize the device [9].
- 8. Before the device can be used, initialization is required to form the conductive filament. A voltage is applied across the RRAM device that allows a CF to form. The forming current affects the thickness and length of the CF which in turn affects the resistance of the device [9].

2.3.5. Defects in RRAM devices

Overview

To be able to characterize and identify defects in RRAM devices we need to known the type of defects that can occur in RRAM devices. Defects are unwanted differences caused by the manufacturing process between the intended design and the actual device. The defects can be linked to a specific manufacturing step. For FEOL the defects can be divided in historical and emerging defects [57]. For the BEOL stage a distinction can be made between defects in the interconnect and the RRAM Device. An overview of the different types of defects is given in Table 2.3.

FEOL defects

The FEOL stage of the fabrication process refers to defects that occur during during the production of the transistor. These defects have been studied extensively. These defects can be classified into two categories: Historical defects and emerging defects [57]. Historical defects refer to variations that have been observed over and analyzed over time. These defects include patterning proximity effects, regions that should by protected by the photoresist are still affected by photons. Line-edge and line-width roughness are variations in the edge and width of the pattern resulting in a rough surface. Polish variations which impact the thickness and uniformity of the layers. Variations in the gate dielectric, which are caused by impurities in the gate oxide of the transistor [57].

Emerging defects are defects that were insignificant in impacting device performance until the dimensions of the device components were reduced. These defects include random dopant fluctuations, annealing effects, strains and metal granularity [57]. These defects have been studied extensively and test methods have been developed to detect and mitigate their impact on device performance.

BEOL defects

In the BEOL stage defects can occur in both the interconnect and the RRAM device. Defects in the interconnect, which are part of the conventional CMOS process, include opens, shorts, line roughness, irregular shapes, big bubbles and small particles [9, 15].

Defects in RRAM devices can be categorized based on their location, such as defects in the top electrode, bottom electrode, and oxide. An overview of the defects and their effects is presented in Table 2.3. For each manufacturing step the corresponding defects are discussed.

- Metallization gives rise to defects part of the conventional CMOS process. These defects in the interconnect include opens, shorts, line roughness, irregular shapes, big bubbles and small particles [9, 15].
- BE patterning is done using physical vapor deposition. As mentioned PVD leaves a rough surface Generally this is done using physical vapor deposition. PVD leaves a rough electrode surface which needs to be planarized. The rough surface leads to surface defects between the oxide and the bottom electrode. This increases the probability of hard oxide breakdown and device variability[9, 15].
- 3. Improper BE planarization will give rise to polish variations which are similar to the variability that can occur during the BE patterning. The under- and over etching of the BE can result in opens, shorts and bridge defects [9, 15].
- 4. Metal-oxide deposition uses ALD to control the thickness of the oxide. Thickness variations in the oxide will affects the the resistance and switching behaviour of the device increasing the device-to-device variability. Additionally, the oxide has a crystalline structure of which the edges may contain grain boundaries. These boundaries have a higher concentration of defects and conductive filaments tend to form along these boundaries [9, 15].
- The addition of the capping layer and TE can give rise to sidewall redeposition creating a path with lower resistance affecting the device behaviour. Additionally, variation in the thickness can significantly reduce the resistance of the device[9, 15].
- 6. TE deposition gives rise to similar defects as the BE patterning. Surface roughness can affect the device behaviour. Additionally, sidewall redeposition can create a leakage path lowering the devices resistance.

- Post-processing methods are applied to finish the RRAM device. The etching can give rise to redeposition of materials and sidewall leakage, affecting device behaviour. Additionally, the RRAM devices on the wafer need to be separated, which can lead to variations in the device's dimensions as well as sidewall redeposition[9, 15].
- The forming current affects the thickness and length of the CF which in turn affects the resistance of the device. Fluctuations in the forming current will increase the device-to-device variability and in extreme cases increase the chance of a hard breakdown[9, 15].

In the BEOL stage defects can occur in both the interconnect and the RRAM device. Defects in the interconnect, which are part of the conventional CMOS process, include opens, shorts, line roughness, irregular shapes, big bubbles and small particles. [15] Defects in RRAM devices can be categorized based on their location, such as defects in the top electrode, bottom electrode, and oxide. An overview of the defects and their effects is presented in Table 2.3. The deposition process of the BE can result in a rough surface on the BE, leading to surface defects between the oxide and the bottom electrode. This increases the probability of hard oxide breakdown and device variability. To address the rough surface, the BE is planarized which can give rise to polish variations [9, 15]. The oxide thickness affects the device resistance and switching behaviour of the device. To achieve tighter control over the oxide thickness, atomic layer deposition is used. The oxide has a crystalline structure of which the edges may contain grain boundaries. These boundaries have a higher concentration of defects and conductive filaments tend to form along these boundaries [21, 32]. The addition of the capping layer and TE will give rise to sidewall redeposition creating a path with lower resistance affecting the device behaviour [9]. The main focus of the work is on the identification and characterization of defects in RRAM devices. Table 2.3 presents an overview of all the defects and the effect on the electrical parameters of the device.

Location	Defect	Defective behaviour	Parameters affected.		
	Contamination	Affects the heat conductance and the quality of the forming process	HRS LRS V _{set} V _{reset}		
	Increased electrode roughness	Affects the contact with the oxide. Which affects the resistance and the chance of a hard breakdown.	HRS LRS V _{set} V _{reset}		
	Poor bonding with contact	Increases the contact and line resistance.	HRS LRS		
BE	Thickness variations	Affects the heat conductance of the device	HRS LRS		
	Over- and under-etching	Results in opens, shorts and bridges. Which affects the line resistance and capacitance	HRS LRS		
	Contamination	Increases the variability of the device; Increased or decreased resistance	HRS LRS V _{set}		
	Variations in the crystalline structure	riations in the crystalline structure Concentration of oxygen vacancy.			
Metal-Oxide	Thickness variations	Affects the switching characteristics of the device. Number of grain boundaries; Concentration of oxygen vacancies	$ \begin{array}{l} HRS \\ LRS \\ V_{set} \\ V_{reset} \\ t_{H \Rightarrow L} \\ t_{L \Rightarrow H} \end{array} $		
	Poor bonding with BE	Affects resistance; Increased set/reset voltage; increased chance of hard breakdown	HRS LRS		
	Sidewall leakage	Increased variability; reduced set voltage and resistance.	HRS V _{set}		
те	Surface roughness	ness Affects resistance; device variability			
16	Poor bonding with capping layer	Affects resistance; device variability	HRS LRS		
	Overforming	Increased variability of HRS. Increased retention.	HRS V _{set} V _{reset}		
Forming	Underforming	Decreased variability of HRS; Lower retention.	HRS V _{set} V _{reset}		

Table 2.3: The effect of manufacturing defects on the device characteristics.

3

State of the art in characterization and diagnosis

3.1. Overview

This chapter present the state of the art in defect characterization and diagnosis. Defect characterization attempts to understand how defects affect the performance and electrical behaviour of the device. Where diagnosis focuses on finding the cause and location of the defect. The methods for defect characterization are divided in three categories: Electrical characterization, memory testing and optical characterization. Electrical characterization focuses on identifying defective devices based on electrical measurements of the RRAM cells. Memory tests focus on fault identification using fault models. Fault models are an abstraction of the defect at an electrical level that mimic the faulty behaviour of the device [13]. Optical characterization uses optical measurements to study the physical characteristics of the device. In some cases this is combined with electrical measurements. The three different techniques for defect characterization are evaluated and compared.

3.2. Electrical characterization

Electrical characterization attempts to characterize RRAM cells using electrical measurements from RRAM cells and arrays. In general there are three different setups to measure RRAM: I-V sweeps and pulsed voltage sweeps(PVS) which can be further divided into current-visible and current blind PVS [19].

3.2.1. I-V sweeps

In an I-V sweep the electrical behaviour of the RRAM device is studied by gradually increased the voltage [19]. The device is connected to a probe station. One of the electrodes is grounded and an increasing voltage is applied to the other electrode [58]. To prevent damaging the device the current is limited to reduce the amount of energy supplied [19, 58]. While gradually increasing the voltage, the current going through the device is measured. In the case of bipolar switching, a positive and negative voltage sweep is done [19]. The positive sweep monitors the SET process, the negative sweep monitors the RESET process. An I-V diagram is generated that can be used as proof-of-concept and to verify that the RRAM cell switches properly [19]. Figure 3.1 shows a typical I-V graph using an I-V sweep. The advantage is that you can guarantee the correct switching of the device as it provides insight into the intrinsic switching characteristics and visualizes one complete switching cycle. The downside is that one I-V sweep takes up to 60 seconds to generate and the method does not match the normal operation conditions [19]. RRAM normally operates using pulsed voltage sweeps. The voltage stress applied during PVS is larger which can affect the endurance of the device. However, because I-V sweeps show the entire switching cycle it is frequently used for the characterization of RRAM [19].



Figure 3.1: Current vs voltage diagram using I-V sweep showing the intrinsic switching characteristics.

3.2.2. Pulsed voltage sweeps

Current-visible PVS

In CV-PVS short voltage pulses larger than >1 V are applied to the device while monitoring the current of the device. One pulse may have a duration on the order of microseconds, which is much faster compared to the 60 seconds needed for a single I-V sweep [19].

The main advantage of CV-PVS is that one cycle only lasts a few microseconds and that the switching characteristics are comparable to how a device usually switches between states. However, PVS is more expensive as the current needs to be measured which is usually done using a parameter analyzer. Furthermore, the intrinsic switching characteristics of the RRAM device can not be studied as you can only retrieve the state of the device before and after switching by applying a small read voltage [19].

Current-blind PVS

CB-PVS does not monitor the current and is the cheapest method to measure RRAM devices. After a specified number of cycles the resistance of the device is measured. The method is fast and cheap but you cannot guarantee that the device has switched each cycle making it unreliable.

The different measurement setups can be used to study the characteristics of RRAM devices. The electrical characteristics of a RRAM device are described using the following metrics: endurance, variability, retention, switching time and energy consumption. These metrics are described using the electrical parameters: $V_{set} V_{reset}$, HRS, LRS, $t_{L \Rightarrow H}$ and $t_{H \Rightarrow L}$. The electrical parameters have already been described in subsection 2.3.3. A brief description of each metric is given here

- The endurance describes the number of times a device can switch between states where the sensing margin between HRS and LRS (*R_{ratio}*) is large enough. After each cycle the *R_{ratio}* is determined by extracting the resistance in HRS and LRS. The common method to visualize the endurance is a HRS and LRS vs. cycle graph. Figure 3.2 shows an endurance diagram. Normally, the number of cycles should be larger to assess the endurance of the device [19].
- The retention describes how long the RRAM cell will preserve its state. This mainly concerns the LRS, as HRS is the natural state of a RRAM cell. This is measured by exposing the device to a constant voltage over time by applying a read voltage of 0.1 V. Which is visualised using an current-time figure.
- The variability describes the device-to-device and cycle-to-cycle variability. This is done using the electrical metrics. These metrics are visualized using a cumulative density graph or histogram. When using I-V sweeps this can also be depicted by showing multiple cycles in a single graph.

- Switching time describes how long the device needs to switch from HRS to LRS and vice versa. This is typically determined using CV-PVS by applying triangular voltage pulses to the device.
- Energy consumption can be calculated by integrated the current and voltage over time.



Figure 3.2: Endurance graph by plotting HRS and LRS for each cycle.

Table 3.1 shows which measurement setup can be used for the different metrics. Overall, CV-PVS and the I-V sweep give the most information and CB-PVS can only be used to determine the endurance of a RRAM cell.

Table 5.1. A companyon between the uncrent measurement setups.	Table 3.1: A comparis	son between the	e different meas	urement setups.
----------------------------------------------------------------	-----------------------	-----------------	------------------	-----------------

Method	Endurance	Variability	Retention	Energy consumption	Performance	Cost	Visual inspection
I-V sweep	Yes	Yes	Yes	Yes	Slow	Normal	Yes
CV-PVS	Yes	Yes	Yes	Yes	Fast	Expensive	No
CB-PVS	Yes	No	No	No	Fastest	Cheap	No

In literature, these metrics have only be used for characterization and not for defect identification of RRAM cells. Zahoor, Zulkifli, and Khanday performs a series of I-V sweeps in which the endurance is used as a metric to analyse the impact of device dimensions [2]. Different cell sizes were used. The endurance of cell sizes with 10 nm,20 nm,30 nm and 40 nm was lower compared to larger devices. Fantini et al. analysed the effect of oxides thickness on the endurance of RRAM cells. Devices with an oxide thickness of 10nm performed better than an oxide thickness of 2nm [20]. Grossi et al. evaluated the impact of grain boundaries using I-V sweeps on array level for both polycrystalline and amorphous HfO₂ [21]. The polycrystalline structures showed higher cell-to-cell variability which is contributed to the effect of the grain boundaries. The effect was illustrated using the cumulative density graphs of V_{set} and V_{Reset} .

In Table 2.3 the defects that can occur during manufacturing and how they impact the electrical parameters of the device have already been discussed. However, no one has yet attempted to use these electrical parameters to identify these defects in RRAM devices.

3.3. Memory tests

Memory tests attempt to identify defective devices using fault models. A fault model is an abstraction of a defect that describes the faulty behaviour at the electrical level. The defective behaviour of a fault is described by the memory operations that sensitizes the fault as well as the actual value stored in the memory after sensitization of the fault [13].

A fault can be described as a fault primitive using the following notation $\langle S/F/R \rangle$ [14, 15, 59, 60]. In [15] this notation was adapted so that faults with similar behaviour are grouped together. The notation is:

- S describes the chain of memory operations required to sensitize the fault. The notation is described as $x_0 O_1 x_1 \cdots O_n x_n$. *x* describes the value that is stored in the cell and *O* describes the type of operation e.g. read or write. E.g. S = 0w1r1 indicates a memory cell which holds a '0' to which a '1' was written so the expected read output is a '1'.
- F described the value that is stored in the memory cell after the sensitizing memory operation. For the current memory types this is either a '0' or a '1'. However, because RRAM is an analog device more states can be defined than '0' and '1'. Besides '0' and '1', RRAM also has an extremely high resistance state (*H*) in which the resistance is larger than HRS, extremely low resistance state (*L*) where the resistance is lower than LRS and an undefined state (*U*) in which the resistance value is between HRS and LRS.
- · R describes the output after a read operation. When there is no read operation

Generally speaking, faults can be divided into two categories: strong and weak faults which can be further divided into Easy-to-Detect (EtD) and Hard-to-Detect (HtD) faults [15]. Strong faults are faults that can be described by a fault primitive e.g. (1W0/1/-). Weak faults are faults that cannot be described by a fault primitive because the cells content remains unaffected. Strong faults that are guaranteed to be caught after a specific memory operation are called EtD. This is not necessarily the case for all weak faults. For example, (0r0/0/?) the read value can either be a '0' or a '1'. These faults together with all strong faults are HtD.

For RRAM a distinction can be made between conventional faults models and unique faults model [14]. Conventional faults are faults that can also be observed in mainstream memories such as SRAM and DRAM. Unique faults are new emerging faults caused by the analog characteristics of the RRAM device. A brief description of the conventional faults found in literature are:

- Stuck-at-Faults are faults where a cell is always in a specific state. E.g. (0/1/-), (1/L/-) are fault primitives belonging to this class. These can be caused by an defect in the word- or bit line. E.g. a short or a missing link in the WL or a defect in the forming of the device [14, 17].
- Transition Fault, the cell fails to switch from HRS to LRS or vice versa after a SET or RESET operation within the allowed time. Examples of FPs belonging to this category are (1w0/1/-), (0w1/H/-). Can be caused by defects in the interconnects and or the transistors. A defect in the capping layer or contacts [14, 17].
- State coupling Fault [61, 62, 63]: Two or more cells are connected via a bridge, bit line or world line. When the state of the addressed cell is altered the state of the neighbouring cell is affected as well.
- Write disturbance fault is caused by a defective transistor on the word line or bit line [61]. During a write operation a cell on the same line as the addressed cell also changes state.
- Read disturb fault [22]: A read fault that due to the small bias current required to read a cell changes the state of the cell.
- Incorrect read fault [64]: The addressed cell contains the correct value before and after the read
 operation but the wrong value is returned. This is mainly caused by resistive defects in the RRAM
 cell.

The unique faults are caused by the additional states present in RRAM. The ones found in literature are:

- Undefined write fault [17]: The cell fails to switch from HRS to LRS or vice versa after a SET or RESET operation and remains in the undefined state. So after a read operation either a '1' or '0' is returned. This is an indication that the bias current in the cell is too low which can be explained by a weak SL line.
- Deep state fault [65]: The resistance is larger (lower) than the boundaries defined for HRS (LRS), which can be a result of overforming [14].
- Unknown read fault [66]: A random value is returned after a read operation. This means that either the sensing margin is too small and LRS and HRS are too close to each other or that the device is in the undefined state.

To identify these faults different testing techniques have been designed that are able to detect and sensitize the conventional and unique faults. Generally march tests or specialized Design-for-testability (DfT) are used for fault detection. A March test is a sequence of read and write operations that are able to sensitize the fault [22]. DfTs add additional hardware to make the manufactured device more easily testable [15]. For example by adding scan chains so that internal states within the circuitry can be extracted. These test methods are widely used within the industry and specialized DfTs and march tests have been developed to catch the conventional and unique faults in RRAM.

The problem is that these fault models cannot be linked back to the specific defect of the RRAM device as multiple defects can result in the same fault. In mainstream memories, defects are modelled using linear resistors [14, 15, 16, 17]. For RRAM and STT-MRAM this has also been the main strategy [67, 68]. However, the fault models based on the linear defect models are not an accurate abstraction of the defect. Fieback et al. created an model for the over-forming defect and compared this to traditional method where a linear resistor is used to mimic the over-forming defect [15]. The results showed that the fault space was completely different and that the linear resistor could not be used to model the over-forming defect. As a result, the yield loss is low because we design tests for the wrong fault primitives leading to test escapes and low quality devices.

Memory tests are used for identifying defective devices. However, the fault models are incomplete and inaccurate. Additionally, these can no longer be linked back to the specific defect that resulted in the fault and cannot be used for defect identification.

3.4. Optical characterization

The most expensive method to identify defects in RRAM devices is optical characterization. Optical characterization uses optical measurements to study the physical characteristics of the device. Different methods have been used to examine RRAM devices for defect characterization. The most common practice is transmission electron microscopy (TEM), atomic force microscopy (AFM), scanning electron microscopy (SEM).

3.4.1. TEM

A focused beam of electrons is transmitted through the specimen to visualize the internal structure of the cells at nanoscale. In RRAM, this has been used to visualize the RRAM structure at both cell and array level. Additionally, this method has been used to study in-situ switching in RRAM cells [26, 27, 28, 29, 69].

At cell level TEM is frequently used to characterize the resistive switching real-time [26, 27, 28, 29]. For in-situ TEM an scanning tunnelling microscopy (STM) tip and holder is used to apply electrical stress to the cell.

Kwon et al. was the first to study the resistive switching using in-situ TEM to study the filamental switching of 3 by $1.5 \,\mu$ m single crystal TiO₂ devices. The conductivity of the filament is described as the extension and retraction of Wadsley defects indicated the migration of oxygen vacancies. When there was no extension there was no set process [26]. However, TEM was insufficient to analyse the complete (Pt/TiO_{2/}TiN) cells. The thickness of the specimen prevented the observation of structural changes. For this scanning transmission electron microscopy (STEM) was used in combination with high-angle

annular dark field (HAADF) imaging. Sun et al., Wang and Wu performed similar measurements but with different device structure: $(Ni/HfO_2/SiOx/n^+)$ and ZrO_2 based RRAM devices [27, 28].

Pey et al. studied the effect of the compliance current on the switching of RRAM cells using insitu TEM. A specimen of 100 nm was processed to generate multiple metal insulator semiconductor structures (Ni/HfO_{2/}SiOx/n⁺) which are isolated using focused ion beam milling. For forming a voltage sweep from 0 V to 5 V is done. Then for different compliance currents the switching behaviour was studied. They found that for compliance currents larger than 10 nA an unique geometric defect was observed due to metal migration from the top electrode [29]. Which could be related to the intermittent undefined state faults described [70] [70].

The downside is that only a thin slice can be used for TEM [26, 27, 28, 29, 30]. The thickness of the device is not representable and can drastically alter the switching statistics of the device. Additionally, RRAM cells using in-situ TEM are studied at cell level and not array level consisting of an additional transistor, word and bit lines. TEM does not incorporate these influences and can only be applied to destructive ex situ imaging experiments [19].

3.4.2. AFM

AFM uses a probe tip which scans over the sample surface. The tip interacts with the atomic forces between the surface and the tip creating a topological map of the surface. Unlike, TEM for AFM the sample thickness does not matter and is in preferred over TEM due to sample thickness. Nandi et al., Charpin-Nicolle et al. uses AFM to measure the electrode roughness of RRAM cells. Both studied the forming voltage as a function of electrode roughness. In which the failure rate of devices increases when the electrode roughness increased [30, 31].

Variations on AFM use a conductive tip (CAFM) so that electrical characterization can be performed in-situ while measuring the surface topology. However these cannot be used for endurance tests, the CAFM tip suffers from lateral thermal drift and the high current density is harmful for the tip. Limiting the number of cycles to up to a 100 [71, 72].

In [32] CAFM is used in-situ to show the impact of grain boundaries and that the resistive switching of RRAM cells is more likely to occur at those boundaries [32]. Hence as the device dimensions decrease the impact of single grains grows larger affecting the device to device variability.

In [73] CAFM is also used to demonstrate the switching of RRAM cells. However, they also used X-ray spectroscopy to show that the oxygen atoms migrate to the anode changing the composition of the NiO film which induced a resistance change [73].

3.4.3. SEM

SEM is mainly used for visualization purposes of RRAM cells and array for visualization of RRAM cells and arrays [74].

The optical techniques are able to identify the process variations which lead to defects in RRAM cells. The physical dimensions of the RRAM device can be measured accurately. Electrode roughness can be monitored which affects the variability and the forming voltage of the device. Additionally, the switching mechanism can be studied real-time. However, due to the high current density, the number of cycles that can be characterized is limited. The tools required is expensive and measurements are time consuming. Furthermore, the optical inspection can only be applied to isolated cells and has not yet been extended for characterization at array-level. The inspection of RRAM devices using optical measurements is time consuming and costly and in some cases even destructive for the RRAM cell [19].

3.5. Conclusion

The three methods for defect characterization have been compared. Optical characterization is the most accurate, but is expensive. The physical examination of the RRAM device makes it easy to detect defects such as electrode roughness which can be studied in-situ. However, due to the high current density, the number of cycles that can be characterized is limited. Optical characterization is mainly used for individual RRAM cells and has not yet been extended to array-level characterization. Additionally, the physical dimensions of the device are not realistic as the thickness of the device is thinner than what is found in practice. The memory tests identify defective devices using fault models. The tests can be linked back to the fault model. However, the fault models in most cases cannot be

linked back to the specific defect. Additionally, most fault models are based on the linear resistance models which does not incorporate the analog behaviour of the device decreasing. Which affects the yield loss as we do test for faults that may not exist and vice versa leading to test escapes. Due to the incorrect, fault models and the inability to link the faults back to Electrical characterization is a cheap defect characterization method. This method can be used to detect faults in devices such as the undefined intermittent-state-fault, but may also be used to identify defects in RRAM at both cell and array-level as we have an basic understanding of how the manufacturing defects affect the electrical behaviour of the device. So far electrical characterization has mainly been used to study the switching characteristics and variability of RRAM devices. However, nobody has attempted to identify defective devices using electrical characterization. In the next chapters, we will evaluate the best method to automatically identify defective RRAM devices using electrical characterization.

4

RRAM data analytics

4.1. Overview

Chapter 3 showed that the electrical characterization of RRAM at cell and array level has not yet been attempted for defect identification. Chapter 4 presents a general characterization of the RRAM data using electrical characterization. First, the structure of the RRAM devices are discussed. This is followed by an evaluation of the different methods to extract the electrical parameters. The electrical parameters and visual inspection of the device are used to manually label defective RRAM devices. This is followed by a complete characterization of the devices which will be used to evaluate the important metrics used to automatically identify defective devices.

4.2. Device structure

The RRAM devices have been fabricated by ST Microelectronics using their 130 nm technology. For the top and bottom electrode TiN is used, for the oxide layer HfO_2 and a Ti capping layer is used. The structure is shown in Figure 4.1. The measurements are performed using a staircase voltage sweep with a step-size of 20 mV ranging from -1.8 V to 1.2 V. The parameters were extracted using the Keysight B1500 semiconductor parameter analyzer. The current flow and the corresponding time is measured across the device. The wafer consists of 49 devices arranged in a 1T-1R crossbar array. For each device 936 IV-sweeps were done.

4.3. Metric extraction technique

This section discusses the extraction methods for the electrical parameters: V_{set} , V_{reset} , $t_{H\rightarrow L}$, $t_{L\rightarrow H}$, R_{set} and R_{reset} . For each parameter the extraction methods are evaluated and compared. The performance of the different extraction methods are evaluated using CDF plots, individual IV plots and summary statistics.

4.3.1. Vset

The set voltage describes the point where the device switches from the high resistance state (HRS) to the low resistance state (LRS). Maldonado et al. describes three numerical methods to extract V_{set} from I-V sweeps[75, 76] of which two are evaluated in this thesis. The method that is not evaluated requires an additional tuning parameter and does not perform as well as the other two methods [75]. The variance of V_{set} is larger and the results are severely affected by the tuning parameter [75]. The tuning parameter will add additional complexity and make comparison to other data sets less reliable. The first method that is evaluated uses the current derivative to determine V_{set} . The second method uses the maximum distance to an imaginary straight line that connects the start and end point of a set curve [75].



Figure 4.1: RRAM device:TiN/10 nm HfO₂/10 nm Ti/TiN

Method 1: Current derivative

The first method proposed by Maldonado et al. uses the current derivative to determine V_{set} [75]. The maximum value of the derivative defines the point where the conductive filament forms the bridge between the two electrodes which characterizes the switch from LRS to HRS. The numerical noise in the measurement data can be high under influence of external fields and temperature fluctuations [75, 76]. Figure 4.2a illustrates the method for the one point derivative. To account for the numerical noise we will need to take the derivative over a larger interval to filter out the noise. Maldonado et al. uses a five point numerical derivative Equation 4.3 to account for the numerical noise [76]. However, we have also included the one-, three- and seven-point stencils to calculate the numerical derivative. The expectation is that the one-point stencil will be less accurate due to the noise in the measurement data. Increasing the number of neighbours will improve the accuracy but the computational cost will increase. The equations for the different stencils are shown in equation 4.1,4.2, 4.3 and 4.4.

$$f'(x) = \frac{f(x+h) - f(x)}{h}$$
(4.1)

$$f'(x) = \frac{f(x+h) - f(x-h)}{2h}$$
(4.2)

$$f'(x) = \frac{f(x-2h) - 8f(x-h) + 8f(x+h) - f(x+2h)}{12h}$$
(4.3)

$$f'(x) = \frac{-2f(x-3h) + 9f(x-2h) - 45f(x-h) + 45f(x+h) - 9f(x+2h) + 2f(x+3h)}{60h}$$
(4.4)

Method 2: Maximum separation to a straight line that connects the start and end point The second method described by Maldonado et al. calculates the orthogonal distance from each data point to an imaginary line. Figure 4.2b illustrates the method. The imaginary line is a line through the origin at 0 V and the end point at 1.2 V [75, 76]. The point that is furthest from the line is defined as V_{set} . At V_{set} the curve visibly bends and increases rapidly. This point can be described as the knee curve [76, 77]. The knee curve is the furthest away from the imaginary line [76].


(a) Illustration of extracting V_{set} using method 1 the current (b) Illustration of extracting V_{set} using method 2 an imaginary line. derivative.

Figure 4.2: Extraction methods for Vset

Conclusion

Table 4.1 shows the mean and standard deviations for the different methods using all cycles of the 49 devices. The standard deviation and average are similar for the different extraction methods. However, Table 4.1 does not provide any information about the distributions and how well the methods performs for the edge cases. The CDF in Figure 4.3 shows that the one-point-stencil defines Vset one time-step earlier. Where the set voltage lies 0.03V higher for the method that calculates the maximum distance to the line. Overall, the five methods perform similar and in most cases accurately define V_{set} . The difference between the stencils is negligible. As a result, method 1 is referred to as the method which uses the derivative to determine Vset and method 2 uses the maximum distance to an imaginary line to determine V_{set} . Figure 4.4 shows three IV plots in which the edge cases can be observed for the extraction of V_{set}. In 2.5% of the cycles, method 2 finds a V_{set} that differs at least 0.08 mV. Overall, the switch from HRS to LRS is instant as can be seen in Figure 4.4a. However, in some cases the transitioning from HRS to LRS happens in two steps as is illustrated in Figure 4.4b. Method 1 is more susceptible to the second slope as it is based on the maximum derivative and as a result does not determine V_{set} accurately, while method 2 does work. In only a few cases method 2 does not extract V_{set} correctly. When the transitioning from HRS to LRS is a gradual process method 1 performs better than method 2 Figure 4.4c. This occurs in a few cases (< 0.1%) and is acceptable. As a result, method 2 will be used to extract Vset

Table 4.1: V_{set} mean and standard deviation for the different extraction methods

Method	Maximum distance	One-point	Three-point	Five-point	Seven-point
Mean	0.8083	0.7601	0.7771	0.7763	0.7765
σ	0.1202	0.1121	0.1120	0.1122	0.1124

4.3.2. Reset voltage

Maldonado et al. describes five methods to extract V_{reset} [75]. In this work only two of the five methods are implemented and evaluated. Method one is similar to method one of V_{set} . The method determines the minimum current derivative of the reset curve. The second method that is implemented finds the maximum current. The other methods are not implemented because the methods add unnecessary complexity or are less reliable. One of the methods that is not implemented requires an additional tuning parameter. Another method uses the first point at which the current starts to decrease. However, due to the electrical noise we have in our measurement setup this method has not been implemented transforms the I-V curve to the charge flux domain [75]. This method has not been implemented due to the additional complexity and is time consuming as all the data has to be transformed first. The results in [75] show that the difference between method one and method five are similar. Method one and method two are discussed in more detail as they are used in our work for the extraction of V_{reset}



Figure 4.3: Cumulative density plot (CDF) of the five different methods to extract V_{set}.



(a) IV plot of 1T-1R device WL:0 BL:1 cy-(b) IV plot of 1T-1R device WL:2 BL:0 cy- (c) IV plot of 1T-1R device WL:6 BL:1 cycle:115 cle:217.

Figure 4.4: IV sweep of three different 1T-1R devices. Figure 4.4a in which both method 1 and method 2 extract V_{set} accurately. Figure 4.4b where only method 1 works. Figure 4.4c in which only method 2 works.

Method 1: Minimum current derivative

The minimum current derivative described by Maldonado et al. is similar to method one of V_{set} [75]. Figure 4.5a illustrates the method. The figure shows the reset process as well as the corresponding derivative. The process from LRS to HRS is opposite from V_{set} . So instead of finding the maximum current derivative we know need to determine the minimum current derivative. Unlike with V_{set} the rupture of the conductive filament is not as smooth as the formation and is a more gradual process which contains more measurement noise. As a result, the three-point or seven-point stencil might be more suitable to determine the derivative. The results are shown in Table 4.2 and Figure 4.6.

Method 2: Maximum current

The maximum current is used more frequently to determine V_{reset} [75, 76, 78, 79]. Figure 4.5b illustrates the method. The red point is the point where the maximum current is. As the voltage decreases to -1.8Vthe current running through the device will increase. As soon as the filament breaks the resistance will increase and the current will decrease. This is the start of the transitioning from LRS to HRS and defines V_{reset} . However, due to hysteresis the reset point can be classified at the end of the curve. [75]. To account for this the search space for V_{reset} should be constricted to -0.2 and -1.5 volt. These values are based on Maldonado et al. [75]. Maldonado et al. checked the interval between 30% and 80% of the voltage range [75]. However, in some cases the device had already switched. As a result the interval was changed to 10% and 80% of the voltage range with -0.2 and -1.5 volt.

Conclusion

The average V_{reset} of method 2 varies with more than 0.4 V compared to method 1 which uses the current derivative (Table 4.2). This is already a clear indicator that the two methods are not coherent. To assess which method performs better the CDF is shown in Figure 4.6. Here we see that CDF of the numerical derivative is much more spread out compared to method 2. Investigating the individual IV plots in Figure 4.7 we can see that method 2 can accurately extract V_{reset} while method 1 does not.







(b) Illustration of extracting Vreset using the maximum current.

Figure 4.5: Extraction methods for Vreset

This is due to the fact that our measurement data contains a lot of electrical noise. The switch from LRS to HRS takes longer and consists of multiple steep slopes. This makes it difficult to determine V_{reset} using the minimum derivative.

Table 4.2: Vreset: mean and standard deviation for the different extraction methods

Method	Maximum current	One-point	Three-point	Five-point	Seven-point
Mean (V)	-0.7166	-1.1297	-1.1505	-1.1227	-1.1183
σ (V)	0.0888	0.2984	0.3009	0.2986	0.2991

Figure 4.6: Cumulative density plot (CDF) of the five different methods to extract Vreset.



4.3.3. Resistance

The universal method to extract the HRS and LRS is described by Lanza et al. [19]. The common practice to extract the HRS and LRS is to extract the current at a read voltage of $\pm 0.1 \text{ V}$ -0.3 V [19]. Figure 4.8 shows the boxplots of LRS and HRS for different read voltages. In LRS averages of the different read voltages are similar with 9918 Ω , 9964 Ω and 9976 Ω , respectively. The median is slightly lower for a read voltage of 0.3V. For HRS we find that the difference in resistance is larger for different read voltages. With 523982Ω , 462035Ω and 415072Ω . While the median and standard deviation decreases as well. The R_{ratio} at a read voltage of 0.1V is 21.8% higher compared to when a read voltage of 0.3 Volt is used. This can be explained by the fact that in some cases the device is already switching from HRS to LRS at 0.3V. Figure 4.9 shows an I-V plot of a device with a V_{set} of 0.3 Volt. For this reason we from now on use a read voltage of 0.1 Volt.



Figure 4.7: IV sweep of two different cycles. Figure 4.7a in which both method 1 and method 2 extract V_{reset} accurately. Figure 4.4b where only method 2 works.





(a) LRS distribution for read voltage at 0.1,0.2 and 0.3 Volt

(b) HRS distribution for read voltage at 0.1, 0.2 and 0.3 Volt

Figure 4.8: Boxplot of LRS and HRS for different read voltages. Figure 4.8a shows the LRS distribution and Figure 4.8b the HRS distribution.



Figure 4.9: I-V plot of device with V set of 0.3 Volt

4.3.4. Switching time set

The switching time is only described for PVS in literature [19]. However, in our measurement data we also record the time stamp of each data point. As a result, we can make an estimate for $t_{H\to L}$ if we can determine the start and end point. The starting point is defined by V_{set} which uses the maximum distance to a line. For the end point we have tried two different methods.

In theory, the current derivative will keep increasing after V_{set} until the device has switched from HRS to LRS. So when the current derivative starts to decrease again we have found the end point. However, in practice the noise in our data will create negative slopes during the transitioning from HRS to LRS making this method unreliable and inaccurate.

The second method uses an additional parameter α . The resistance at V_{set} and the minimum resistance is calculated. The end point is then defined using Equation 4.5. An universal threshold value based on the average of all devices or only one device will not find an end point because the maximum resistance is too high, even when the R_{Ratio} is good enough. As a result, the resistance is based on the minimum resistance of each cycle.

$$f'(x) = \alpha R_{min} + R_{min} \tag{4.5}$$

4.3.5. Switching time reset

For the reset the same methods are implemented. Only the tuning parameters are different. The reset voltage is based on the maximum current which is described in subsection 4.3.2. The end point is based using the tuning parameter β and the maximum resistance. Equation 4.6 shows the corresponding equation.

$$f'(x) = \beta R_{max} \tag{4.6}$$

4.3.6. Energy consumption

The energy consumption can be calculated by integrating the current traces over time. According to Lanza et al. depending on the materials used the set and reset energy are similar. To extract the energy consumption from the data we use Equation 4.7. After each time step the current is multiplied with the voltage to get the power generated. This is multiplied by the time to obtain the energy consumed.

$$E = \sum_{i=1}^{n} V_i I_i (t_i - t_{i-1})$$
(4.7)

4.3.7. Endurance

The endurance can only be measured by switching at least 10^6 cycles or by exposing the devices to stress by increasing the temperature. Since the measurement of our data only contains 1000 cycles for each device we cannot use the endurance as a metric.

4.3.8. Variability

The variability is studied using the electrical parameters that have already been extracted. Rather a figure of merit than an electrical parameters. Can be visualised by plotting all the IV cycles in one plot and showing the median for each device [19]. Additionally, cumulative density plots of the electrical parameters can be shown together with histograms to illustrate the cycle-to-cycle variability.

4.4. Nominal device

Before we implement the statistical analysis and machine learning algorithms to automatically identify defective RRAM devices we have to show that the nominal specifications defined in Table 4.3 by ST Microelectronics are insufficient to identify defective RRAM devices. From each device, using cycle 200 up to cycle 936, the *LRS* and *HRS* are extracted using a read voltage of 0.1 V as is described in section 4.3. After the extraction of *LRS* and *HRS* the R_{ratio} can be calculated for each cycle. These values are compared to the boundary values described in Table 4.3. The results are shown in Figure 4.10.

Figure 4.10 shows that only 2 devices pass the hardware specifications. The numbers in the figure indicate the number of defective cycles that were found for that device. While this can be possible it

Parameter	Lower bound	Upper bound
LRS	500Ω	20000Ω
HRS	90000Ω	2000000Ω
Rratio	5	400

Table 4.3: Hardware specification defined by ST Microelectronics.

seems unlikely that from the wafer only 4% of the devices are functional. Twelve devices have less than five defective cycles. The defective cycles need to be inspected to visually to determine whether the cycles are classified correctly.



Figure 4.10: Number of defects per device

Figure 4.11 shows three cycles that have been wrongly labelled using the specifications defined by ST Microelectronics. Figure 4.11a meets the hardware specifications. However, it is evident that the device does not switch from HRS to LRS in the correct way and contains an intermittent undefined state-fault [70]. The switch should be immediate, while here the current is gradually increasing. Additionally, Figure 4.11b and Figure 4.11c do not meet the hardware specifications because the resistance in HRS is too low with $64\,000\,\Omega$ and $88\,723\,\Omega$. Visual inspection of the cycles does not reveal any abnormalities. Due to these two cycles the device on wordline 0 and bitline 3 is wrongly classified as defective.



Figure 4.11: False positives and false negatives using the hardware specifications to identify defective devices.

The visual inspection of the I-V graphs shows that the specifications defined by ST Microelectronics for a nominal device are insufficient to identify defective RRAM devices resulting in false positives as well as false negatives. In chapter 6 the statistical methods and machine learning algorithms will be implemented to automatically identify defective RRAM devices. To evaluate the performance of the automatic defect identification methods we will first propose a method to manually label RRAM devices.



Figure 4.12: Endurance graph showing the HRS and LRS for each cycle

4.5. Labelling of the data

The hardware specifications defined in section 4.4 do not accurately identify defective devices. Additionally, we do not know the the optimal number of cycles required to reliably identify defects. To assess the performance of the hardware specifications, as well as the algorithms used in chapter 6 we will propose a method to manually label RRAM cycles. The first 200 cycles are not taken into consideration because erratic resistance states are shown. Figure 4.12 shows the resistance in the *HRS* and *LRS*. The figure illustrates that the first 200 cycles should not be taken into consideration.

From the 736 cycles of each device the sample size is determined using Equation 4.8, where n is the sample size, z is the z-score, e is the error margin, p the variance and N the population size. Assuming a confidence interval of 95%, margin of error of 5%, variance of 50% and a population size of 736 we need to label 253 cycles for each device to accurately determine whether the device contains defects.

$$n = \frac{\frac{z^2 p(1-p)}{e^2}}{\frac{z^2 p(1-p)}{e^2 N}}$$
(4.8)

For each device 253 cycles are randomly selected. The IV plot along with the electrical characteristics of the corresponding cycle are visualized. The cycle is then classified as either defective, functional or unknown based on the following criteria.

For the criteria the specifications from ST Microelectronics are taken into consideration. However, as was shown in section 4.4 using Figure 4.11 the specifications do not accurately identify defective devices. As a result, additional criteria are introduced to reduce the number of false positives. These criteria are based on the visual inspection of defective cycles and what we found in literature. Figure 4.11a suffers from intermittent undefined state faults and does not show a instant V_{set} [70] but according to the specifications of a nominal devices is labelled functional.

The criteria for a functional cycle are as follows.

- 1. The R ratio is between 5 and 400
- 2. LRS is between $2\,k\Omega$ and $20\,k\Omega$
- 3. HRS is above $90 \text{ k}\Omega$
- 4. The switching of V_{set} is instant as illustrated by Figure 4.11b.
- 5. No intermittent undefined state-fault [70]
- 6. V_{set} is between 0.4 V and 1.0 V
- 7. V_{reset} is between -0.4 V and -1.2 V

Additionally, a cycle is considered defective when one of the unique faults is observed:

- 1. Intermittent undefined state fault
- 2. V_{set} increases gradually as illustrated by Figure 4.11a.
- 3. No switching behaviour observed for V_{set} or V_{reset} .

The cycles that do not fall in the defective or functional criteria are assigned to the third category which specifies the cycles that need further investigation to assess the performance. Figure 4.13a is an example of a I-V graph that is labelled as to be determined. While the device meets the specifications the reset process is suffering from over-reset [80]. While this has not been linked yet to a specific defect it is not nominal device behaviour. Additionally, the criteria of an instant V_{set} is based on the visual inspection of I-V graphs. Figure 4.13b is an example in which the switching of V_{set} is not instant and consists of two steps. While this does not correspond with nominal behaviour we also do not consider the device to be defective. As a result, the device remains unlabelled.





Figure 4.14 shows the results for each device. The figure shows for each device the percentage of cycles classified as functional, defective and those of which we are not certain. In some cases the percentage of correct cycles are high, but a few defective cycles are detected. For *WL0BL6*, *WL1BL0*, *WL1BL1* 15% of the cycles are defective applying the manual labels. *WL0BL5* contains one defective cycle



Figure 4.14: Overview of the defective, functional and healthy cycles for each device. For each device 236 cycles have been labelled.

Figure 4.15 shows the number of defective cycles for each device in a heatmap. Here we see that 12% of the devices do not show defective behaviour assuming that the cycles which have not been classified are also functional. 30% of the devices have less than 5 defective cycles which is $\approx 2\%$ of the cycles. A device is considered defective when it has more than five defective cycle. This tolerance is to account for human error and the inherent variability of the device. Based on the results of the manual labelling 20 devices are considered to be functional and 29 devices are defective.



Figure 4.15: Heatmap which shows the number of defective cycles for each device for the manual labelling.

The results of the manual labelling is for the characterization of the devices, used to train the supervised learning algorithms and to evaluate the performance of the classification algorithms used in chapter 6.

4.6. Characteristics of RRAM device

4.6.1. Overview

This chapter present the characteristics of the RRAM devices. The characterization is required to set up the defect criteria and choose the correct parameters for the classification algorithms in chapter 6. For each device the electrical parameters are characterized. The analysis is used to determine if they can be used to identify defects in RRAM devices. The RRAM devices are characterized using V_{set} , V_{reset} , HRS and LRS

4.6.2. V_{set}

Figure 4.16 compares the devices labelled as defective and functional based on the manual labelling method from section 4.5. The figure shows that on average defective devices have a lower set voltage, 0.84 V versus 0.78 V and a larger standard deviation of 0.133 V compared to 0.103 V for functional devices. However, the overlap between defective and functional cycles is too large to identify defective devices using V_{set} .

The variability of RRAM devices is studied using cumulative density plots and histograms [19]. Figure 4.17 shows the cumulative density plot and Figure 4.17b shows the histogram of V_{set} for functional and defective cycles. From this can be concluded that the spread of the defective cycles is larger and that the V_{set} is lower for defective cycles. The histogram also shows that the mode of the curve for defective cycles is lower and that the variability of defective RRAM devices is larger.

Figure 4.18 shows the results for each individual device in which the defective and functional devices are labelled. The figure confirms that the cycle-to-cycle variability is lower and set voltage is higher for functional RRAM cells. However, a low V_{set} does not guarantee that the device is defective. It is not possible to distinguish device WL3BL5 from the other functional devices using Figure 4.18. While the devices with the lowest V_{set} are all defective. Based on these results it is not possible to only use V_{set} to identify defective RRAM devices.



Figure 4.16: Device-to-device variability for Vset



(a) Cumulative density plot of V_{set} for functional and defective cycles. (b) His

(b) Histogram of functional and defective cycles for V_{set}

Figure 4.17: CDF and histogram of V_{set} of functional and defective cycles



Figure 4.18: Device-to-device variability for Vset

4.6.3. V_{reset}

For V_{reset} a similar methodology is applied for the characterization. Figure 4.19 shows V_{reset} for the functional and defective devices. The spread is comparable which is confirmed by the standard deviation with 0.081 V for the functional devices 0.094 V for defective RRAM cells. The main difference is that V_{reset} has more outliers that fall outside the interquartile range (IQR). Figure 4.21 shows that the defective devices has much more outliers. These outliers could be used to identify defective RRAM devices.



Figure 4.19: Functional vs defective devices for Vreset

Figure 4.20 shows the CDF and histogram of the defective and functional devices. Figure 4.6 confirms that the average and spread are similar. Whilst Figure 4.20b shows that the distribution of the defective devices is right skewed, which is confirmed by the surplus of outliers for the defective devices in Figure 4.19. Analysis of the individual devices in Figure 4.21 reveals that the excessive number of outliers for defective RRAM devices can be used as a metric.



(a) Cumulative density plot of V_{reset} for functional and defective cycles.

(b) Histogram of functional and defective cycles for V_{reset}

Figure 4.20: CDF and histogram of V_{reset} of functional and defective cycles

4.6.4. LRS

Figure 4.22 shows the functional versus the defective devices. The average resistance for the functional group is 9319Ω compared to 9920Ω for the defective group. The standard deviation is much lower 1077Ω compared to 25452Ω . The large difference in standard deviation is explained by the cycles that did not switch from HRS to LRS.

Figure 4.23 shows the resistance distribution for each device. From this can be concluded that the



Figure 4.21: Device-to-device variability for V_{reset}



Figure 4.22: LRS functional and defective group

data distribution is wider for the defective devices. Based on the figure it is not evident which devices are defective and which are functional.



Figure 4.23: Device-to-device variability for LRS

4.6.5. HRS

Figure 4.24 shows the defective devices for HRS. The functional cycles have a median of $355\,682\,\Omega$ and standard deviation of $510\,923\,\Omega$. The defective cycles have a median of $225\,199\,\Omega$ and standard deviation of $527\,517\,\Omega$. The boxplot shows that the distribution of the defective cycles is tighter than that of the functional cycles and that overall the defective cycles will have a lower *HRS*.



Figure 4.24: Device-to-device variability for HRS

Figure 4.25 shows the distribution for each device. From the results it is difficult to conclude how this metric can be used to identify defective devices.



Figure 4.25: Device-to-device variability for HRS

4.7. Conclusion

Based on the characterization of RRAM four metrics will be used to evaluate for the statistical analysis of RRAM devices. These are *HRS*, *LRS*, V_{reset} and V_{set} . The extraction of the switching time is not robust enough to use for the defect identification of RRAM devices.

5

Defect identification methodologies

5.1. Overview

Chapter 4 characterized the RRAM devices, which showed that the values of the electrical parameters in defective devices differed from functional RRAM devices. This chapter describes the different methods to automatically classify RRAM devices. The first part of the chapter focuses on statistical methods to identify defective devices. The second part of this chapter focuses on machine learning techniques for defect classification.

5.2. Statistical methods for defect identification

The hardware specifications described in the previous chapter do not accurately identify defective devices. Two statistical methods have been applied. Method one uses the interquartile range (IQR) to determine new decision boundaries for each of the electrical parameters. The second method uses the functional cycles from the manually labelled data to design a model I-V curve. Using the dynamic time algorithm (DTW) and the Euclidean distance as a metric the cycles are compared to the model I-V curve which is linked back to each device. This chapter describes the method and in chapter 6 the results are discussed.

5.2.1. IQR range

The interquartile ranges provides information about the dispersion of the dataset [81]. The IQR is defined by the difference between the 75th (Q_3) and 25th (Q_1) percentiles of the data. The equation to determine the IQR is given in Equation 5.1.

$$IQR = \frac{Q_3 - Q_1}{2}$$
(5.1)

The IQR can be used as an outlier detection technique. The method uses the median to determine the cut-off and as a result is not sensitive to extreme outliers from defective cycles. The upper bound and lower bound is established by Equation 5.2

$$upper_bound = Q_3 + 1.5IQR$$

$$lower_bound = Q_1 - 1.5IQR$$
(5.2)

 Q_1 and Q_3 are the 25th and 75th percentiles. The IQR is described in Equation 5.1. Traditionally, 1.5 is used since this corresponds to the 3σ used in traditional statistical tests and comprises about 1% of the measurement data. The IQR is used as an outlier technique on the following metrics: V_{set} , V_{reset} , *HRS*, *LRS* and the R_{ratio}

For each metric the upper- and lower bound is determined. This is applied to each cycle which is linked back to the device. The results are discussed in chapter 6

5.2.2. Model I-V curve

The method is a distance based metric based on the difference between an ideal I-V plot and the actual data. The functional cycles from the manual labelling in section 4.5 are combined into one single I-V curve. The result is depicted in Figure 5.1.



Figure 5.1: I-V curve from functional cycles averaged into one individual cycle

Two distance metrics are used to compare the I-V cycles to the model I-V curve: Euclidean distance and dynamic time warping (DTW). Figure 5.2 illustrates the difference between the two algorithms. The Euclidean distance bases the similarity of two I-V plots by adding the straight line distances between the corresponding data points. DTW contains an additional step. DTW first aligns the two data series so that the Euclidean distance between the two series is minimized after which the Euclidean distance is used to calculate the similarity between the aligned points.



Figure 5.2: Euclidean and dynamic time warping algorithm [82]

DTW is computationally expensive and time consuming. As a result, the data is processed in two ways. The first method averages the cycles from each device into a single I-V plot. The I-V plot is compared to the model I-V curve using DTW and the Euclidean distance. This way only 49 cycles have to be processed. The other method is to calculate the distance of each cycle using the two algorithms with respect to the model I-V curve. Then for each device the values of the individual cycles are summed to represent a single value. This value is used to determine whether a device is defective or functional.

5.3. Machine learning methods

For machine learning three methods have been used for the classification of RRAM devices of which one unsupervised and two supervised machine learning algorithms. K-Means is an unsupervised clustering algorithm that attempts to group data based on the distance between the data points. K-Nearest Neighbour (KNN) is a supervised learning data that assigns new data points to the class of its nearest neighbours in the feature space. The third method applies the convolutional neural network (CNN) as described by [83] to identify defective devices. Before the machine learning algorithms can be applied the data is preprocessed using the principal component analysis.

5.3.1. Input data

For the machine learning algorithms we use the measurement data of the RRAM devices in three different methods. The measurement data is transformed to three different input sets:

- 1. Input 1 is the raw measurement data as input as described in section 4.2. A staircase voltage sweep with a step-size of 20 mV ranging from -1.8 V to 1.2 V.
- 2. Input 2 consists of the electrical parameters used in the statistical analysis. These are *V*_{set}, *V*_{reset}, *HRS* and *LRS*. The advantage is that the input size is much smaller and it is easier to determine why the device is defective.
- 3. The third input stores the raw measurement data of each cycle as an image. The x-axis ranges from -1.8 V to 1.2 V and the y-axis ranges from -170 µA to 170 µA. This method is mainly used for the CNN algorithm.

Input 1 is used so that the algorithms might recognize difference between defective and functional devices that we have not yet discovered. Input 2 is to see if the current metrics that we use in literature are valid to characterize RRAM devices. Input 3 is mainly used for the CNN as the algorithm is mainly applied on images.

Principal component analysis

K-NN and K-Means do not work well for high dimensional data [84]. As a result, a principal component analysis is applied to the three datasets to reduce the number of dimensions. This section describes the concept of the principal component analysis and how it is applied to the input data.

The principal component analysis reduces the number of dimensions by creating new variables (Principal components) which are a linear combination of the original variables [85, 86, 87]. The principal components are the eigenvectors of the covariance matrix. The first principal component (PC) maximizes the variance of the data and contains the most information. The second PC is an orthogonal vector on the first PC and describes a percentage of the variance. The result is a new coordinate system that describes the variance of the data in the most efficient way [85, 86, 87]. The python library scikit-learn has a function that can apply PCA to the input data [88].

The PCA is applied to all three input data sets. The number of principal components that is used for the K-Means and K-NN clustering is determined by the percentage of variance explained by the corresponding PCs. This value is set at 95%. For each input the results are shown for the functional and defective cycles of the manually labelled data to see if a difference can be observed between defective and functional cycles. Additionally, the PCA points of the cycles are summed for each device and labelled using the results from the manual labelling of the device to determine whether groups can be distinguished.

5.3.2. Unsupervised method 1: K-means

After PCA, K-Means is one of the methods used to classify the data points from the PCA. The advantage of K-means is that the method works for large datasets, converges quickly and is relatively simple to implement. Additionally, the data does not need to be labelled in advance. The disadvantage is that the clustering method, similar to all clustering methods, is sensitive to outliers.

The K-means algorithm works the following way and requires the number of classes (K) to be known in advance.

1. For each class a random centroid is selected.

- 2. For each data point PCA the distance to each centroid is calculated using the Euclidean distance. The data point is assigned to the centroid that is closest to the data point.
- 3. The centroid are redefined by taking the average of all data points that are in the same cluster.
- 4. Repeat the previous two steps until the convergence criteria is met.

Due to the simplicity of the algorithm the K-Means clustering method is applied to the PCA of all the inputs. The results from the K-Means algorithm are compared to the predefined labels of the cycles and devices to evaluate the performance of the algorithm.

5.3.3. Supervised learning method 1: K-NN

Overall unsupervised learning algorithms are less accurate and since we now have information about which cycles are defective and functional we can also use supervised learning algorithms. The advantage of K-NN is that the algorithm is fast and does not require any training. The labelled data of the functional and defective cycles are stored and based on a similar principle as K-Means. The difference is that with K-NN the clusters are already defined by the labelled data. The algorithm for a new data point consists of the following steps.

- 1. The distance of the new data point to the labelled data points in the training set is calculated
- 2. The number data points K that are nearest to the new data point are determined.
- 3. The new data point is assigned the label that occurs most frequent among the K neighbours.

5.3.4. Supervised learning method 2: Convolutional Neural Network

The CNN architecture is copied from Kocak, Mitard, and Naskali uses the architecture as a pass/fail check in the quality control of FETs [83]. The architecture is used as a baseline for the RRAM devices to use as a pass fail check. The CNN is trained using the labelled data.



Figure 5.3: Architecture of CNN [83]



Results

This chapter provides the results from the different classification algorithms described in chapter 5.

6.1. Statistical analysis for defect identification

6.1.1. Results: Statistical analysis

Instead of the hardware specifications we propose a more statistical approach where the boundaries are defined using the interquartile range. For each electrical parameter we have determined the upper and lower limit using Equation 5.2. The limits for each parameter are given in Table 6.1. The table shows each metric and the corresponding upper and lower bound that was determined using Equation 5.2.

Metric	Lower bound	Upper bound
V _{set}	0.48 V	1.12 V
V _{reset}	-0.91 V	-0.51 V
LRS	6979.80 Ω	12 944.58 Ω
HRS	10.73 log(Ω)	14.80 Log(Ω)
$Log(R_{Ratio})$	1.39	5.69

Table 6.1: Upper and lower boundary for each metric using the IQR.

For each metric the performance is evaluated. Figure 6.1, Figure 6.2, Figure 6.3, Figure 6.4 and Figure 6.5 show the results for V_{set} , V_{reset} , *LRS*, *HRS* and the R_{Ratio} . The figures show a heatmap in which the result of the algorithm is compared with the corresponding labels. The colors represent the false positives, false negatives and the devices that have been classified correctly. The numbers in each box represent the number of defective cycles the algorithm found for that device.

Figure 6.1a shows the devices that have been labelled as defective because V_{set} was below the threshold of 0.48 V. In total the metric as classifier consists of 9 false negatives and 9 false positives which is a accuracy of 62%. Figure 6.1b shows the devices of which the upper limit of 1.12 V for V_{set} is exceeded. 15 devices are wrongly labelled as defective and 9 defective devices are not identified by the upper threshold of V_{set} resulting in an accuracy of 48%.

Similar for V_{reset} , Figure 6.2a shows the number of devices detected as false positive, false negative and correctly classified for the lower threshold and Figure 6.2b for the upper threshold. The lower threshold identifies 4 devices as functional of which 1 device is a false positive and 17 out of 20 functional devices are wrongly classified as defective. Figure 6.2b miss-classifies 7 devices as defective, 25 devices as functional and 4 true defective devices are found.

Figure 6.3a shows the result for the lower threshold in *LRS*. The lower threshold of the hardware specification of 500Ω is too strict. The threshold of 6979Ω determined by the IQR range correctly identifies 4 defective devices. The other devices are all considered functional. Figure 6.3b find 11 false negatives and 2 false positives. The other 36 devices are correctly classified as defective or functional.

Figure 6.4a shows the results for the lower threshold of the logarithmic value of *HRS*. The method identifies the defective devices on word line 3 and 4 and contains 3 false positives. Figure 6.4b does not work. Seven device are correctly identified as defective but also contains seven false negatives.



(a) Identification of defective RRAM devices using the lower bound of V_{set} .

(b) Identification of defective RRAM devices using the upper bound of V_{set} .



Figure 6.1: Defect identification of RRAM devices using Vset



(a) Identification of defective RRAM devices using the lower bound of $$V_{reset}$.$

(b) Identification of defective RRAM devices using the upper bound of $$V_{reset}$.$



Figure 6.2: Defect identification of RRAM devices using Vreset

(a) Identification of defective RRAM devices using the lower bound of LRS. (b) Identification of defective RRAM devices using the upper bound of LRS. LRS.

Figure 6.3: Defect identification of RRAM devices using LRS



(a) Identification of defective RRAM devices using the lower bound of (HRS.



Figure 6.4: Defect identification of RRAM devices using HRS

The logarithm of the R_{ratio} in Figure 6.5 is an extension of *LRS* and *HRS*. The results of lower threshold of the R_{ratio} shown in Figure 6.5a are similar to the results of the lower threshold of the *LRS* and *HRS* metric. The lower bound of the R_{ratio} performs slightly better as only 2 false positives are found. The results of upper threshold of the R_{ratio} are identical to that of the upper threshold of *HRS* and is not a good metric to use as it is to identify defective devices.



(a) Cumulative density plot of V_{reset} for functional and defective cycles.



(b) Histogram of functional and defective cycles for Vreset



The statistical analysis is a combination of different metrics that are analysed to identify defective devices. As a result, devices that are wrongly classified as defective (false negatives) are more important than false positives. However, non of the metrics are able to identify devices without identifying multiple false positives.

6.1.2. Results: Model I-V curve

The golden I-V is a similarity metric that is used to measure the similarity between two RRAM devices. A model I-V curve is generated by averaging the cycles from functional devices as was shown in Figure 5.1, which is compared to the test data using dynamic time warping and the euclidean distance as metric.

Figure 6.6 and Figure 6.7 applies the two algorithms on the averaged cycle for each RRAM device. Figure 6.8 and Figure 6.9 first applies the algorithm to each cycle after which the scores of each cycle are added for each device. The labels from section 4.5 as well as a decision boundary that is optimized for the least number of false negatives (FN) are added to each figure .

Figure 6.6 shows that the dynamic time warping algorithm does not perform well to classify defective



Figure 6.6: Dynamic time warping algorithm applied to each individual cycle and linked back to each device.

devices. Figure 6.10 shows the performance of the algorithm for each decision boundary. The score is based on the percentage of devices that are correctly classified. The highest performance is 69% when the decision boundary is set at 0.14. The decision boundary Figure 6.6 displays the threshold for the least number of FN. Based on this 6 of the 29 devices can be identified as defective using this algorithm.



Figure 6.7: Euclidean distance applied to each individual cycle and linked back to the corresponding device.

Figure 6.7 scores better compared to the dynamic time algorithm applied in Figure 6.6. The highest score is 71% with a threshold of 0.21 (Figure 6.10). If the decision boundary is optimized for the least number of FN 14 defective devices can be identified.

The dynamic time algorithm applied to each individual cycle performs the worst. The highest percentage of devices correctly classified is 67% at a threshold of 0.15. Figure 6.8 shows that 4 of the 49 devices can be correctly identified as defective focusing on the least number of false negatives. However, in practice without the manual labels as reference it will be challenging to determine the boundary as the margin between the functional device (W5B0) and the defective devices is small. Figure 6.9 shows the distance metric for each individual device of the averaged cycles. Optimizing for the number of FN, we can correctly identify 20 from the 29 defective devices which results in an accuracy of 69%. The Euclidean distance as similarity measure for each cycle performs the best. Figure 6.10 shows that 81% of the devices can be correctly classified at a threshold of 0.22.

Based on Figure 6.10 the performance of the euclidean distance as a metric applied to each individual cycle performs the best with a performance of 81%. The other three methods perform similar with a performance of $\approx 70\%$. If we evaluated the score based on the number of false positives the euclidean distance performs better with an accuracy of 48% and 69% for the averaged and not averaged data. Where dynamic time warping scores 21% and 14%, respectively.



Figure 6.8: Dynamic time warping algorithm applied to averaged cycle from each device



Figure 6.9: Euclidean distance applied to averaged cycle from each device



Figure 6.10: Performance of the applied algorithms for difference decision thresholds.

6.2. Unsupervised learning

This section shows the results of the three machine learning algorithms as well as the results from the principal component analysis.

6.2.1. Pre-processing: Principal component analysis

The principal component analysis (PCA) is a preprocessing technique that simplifies complex datasets by reducing the number of dimensions, while keeping the trends and patterns from the data. This method is used as a pre-processing step for the unsupervised learning algorithms as well as the supervised learning algorithms. As explained in chapter 5, PCA has been applied on three different input sets. The raw data, the cycles stored as an image and the extracted metrics.

Figure 6.11 shows the results for each dataset, each figure displays the first three principal components of the PCA analysis. In each figure the labels of section 4.5 are incorporated. Figure 6.11a shows the result in which the measurement data is directly used for PCA. Figure 6.11b shows the result where the PCA is applied on the images of I-V cycles and in Figure 6.11c PCA is applied on the extracted metrics. For each figure two clear clusters can be identified. Figure 6.11a has the functional cycles in the middle of which the defective devices are on the outskirts of the functional cycles. Figure 6.11b contains two groups and the metrics from Figure 6.11c is affected by a few extreme outliers.



(a) Analysis of I-V cycles using raw data as input and manual labels. (b) Analysis of I-V cycles using images of I-V cycles as input and manual labels.



(c) Analysis of I-V cycles using the metrics as input and manual labels.

Figure 6.11: The results of the three different input data using the principal component analysis in which the manual labels from section 4.5 are used as classification

The purpose is to automatically identify defective devices. To link the cycles back to the devices we can either accurately identify defective I-V cycles and link the defective cycles back to the corresponding RRAM device. The other approach would be to add the principal components from each cycle for every device after which a classification algorithm is applied. The results are shown in Figure 6.12. The results of Figure 6.12b and Figure 6.12a look promising as two groups can be identified. For the metrics in Figure 6.12c the two groups seem to have more overlap. While the machine learning algorithms will most likely perform better for Figure 6.12b and Figure 6.12a we will apply the algorithms to all three formats.





(a) Defective and functional devices using PCA of raw data as input including the manual labels.





(c) Defective and functional devices using the metrics as input.

Figure 6.12: PCA plot of the first three principal components using the metrics, raw data and images of I-V cycles as input.

The manual labelling of the RRAM devices is time consuming and prone to human error. As a result, unsupervised learning algorithms are also used to classify RRAM devices. In the next part K-Means, K-NN and CNNs are used to classify the results from the PCA analysis.

6.2.2. K-Means

KMeans is applied on three different input sets. For each input each device is classified using two different techniques. By applying K-means to each individual cycle and linking that back to the device or by applying K-means directly to the devices.

Cycle classification

Figure 6.13 uses the raw data as input and evaluates the performance of the individual cycles. Figure 6.13a shows which cycles the cycles which are wrongly classified as functional, defective and which

are correctly classified. Kmeans does not work well for spectral data in which the functional cycles are surrounded by the defective cycles. Figure 6.13b shows the number of defective cycle for each device. Only the device WL5BL0 does not contain any defective cycles. Furthermore, the device on WL3 and WL4 contain the most number of defective cycles. This could be an indication that these devices contain the same type of defect.



(a) Visualization of the defective and functional cycles after K-means. (b) Number of defective cycles for each device after applying K-Means on raw data as input

Figure 6.13: Result of the Kmeans clustering Figure 6.13a shows the number of correctly classified and wrongly classified cycles. Figure 6.13b shows the number of defective cycles for each device.

Figure 6.14 does not perform well due to the extreme outliers. K-means is not deterministic. However, as Figure 6.14a shows, the outlier is too far apart from the main cluster and is classified as a separate class. As a result, all the cycles are classified as functional devices as is confirmed by Figure 6.14b which shows the defective cycles for each device.



(a) Visualization of the defective and functional cycles after K-means. (b) Number of defective cycles for each device after applying K-means on raw data as input

Figure 6.14: Result of the Kmeans clustering Figure 6.14a

shows the number of correctly classified and wrongly classified cycles. Figure 6.14b shows the number of defective cycles for each device.

Figure 6.15 shows the result of the K-means algorithm in which the I-V cycles are processed as images. The algorithm performs the best for this algorithm as two clear clusters can be identified which can be separated by a straight line. Figure 6.15b shows the number of defective cycles for each device. From this we can see that the functional devices contain few defective cycles, while the defective devices such as WL0BL0 have more than 700 from the 736 cycles that are defective.

Figure 6.16 shows the performance for the different inputs. The green bar presents the percentage of cycles that are classified correctly and the red bar links the classified cycles back to the RRAM devices to determine the percentage of devices that are correctly classified as defective. The classifi-



(a) Visualization of the defective and functional cycles after K-means. (b) Number of defective cycles for each device after applying K-means on raw data as input

Figure 6.15: Result of the Kmeans clustering Figure 6.15a

shows the number of correctly classified and wrongly classified cycles. Figure 6.15b shows the number of defective cycles for each device.

cation algorithm is not 100% accurate as can be seen in Figure 6.16. 90% of the cycles are correctly classified when using the images as input. 70.6% of the cycles are correctly classified using the raw data as input and 63.3% of the cycles are correctly classified using the metrics. To link this back to the devices we have to incorporate a tolerance. This tolerance is based on the visual inspection of the heatmap as well as the percentage of cycles that has been classified correctly. For the images as input the number of defective cycles is set at 25 cycles. For the raw data the allowed number of defective cycles per device is set at 50 and for the metrics the number of defective devices is set at 1. The



Figure 6.16: Performance of K-Means for different input data

percentage of devices is slightly lower with 79.6% compared to 90% of the cycles but still outperform the raw data and metrics as input. When using Kmeans the images should be used as input data.

Device characterization

Applying K-Means on the PCA of the devices results in the following plot for each input set. Figure 6.18, Figure 6.17 and Figure 6.19 show the PCA plot of the first three principal components after K-Means clustering as well as a heatmap to evaluate the performance of the clustering algorithm. Figure 6.17b shows that the clustering algorithm results in two false negatives, twelve false positives and the rest is labelled correctly as functional or defective.



(a) Visualization of the defective and functional cycles after K-means. (b) Visualization of the defective and functional devices after K-means.



Figure 6.18a shows that the clustering algorithm divides the devices over PC 1. As indicated by Figure 6.18b this will lead to 10 false negatives and 6 false positives.



(a) Visualization of the defective and functional cycles after K-means. (b) Visualization of the defective and functional devices after K-means.

Figure 6.18: K-means clustering of PCA using images as input data.

Figure 6.19 seems to perform best for the K-Means algorithm. In total two devices are wrongly classified as functional and 7 devices are wrongly labelled as functional.

Figure 6.20 shows the performance of each input set using K-Means on the PCA results of each device. The algorithm classifies 81.6% using images as input data, 71.6% when the raw data is used as input and 67.3% when the metrics are used as input. The percentage of correctly classified devices is lower for the metrics compared to the raw data. As section 4.4 showed the predefined criteria are too strict and the number of false negatives is much larger. However, the number of false positives is 50% lower. The results need to be further analyzed to determine whether we need to incorporate additional metrics such as the switching time or perhaps some of the metrics are redundant and need to be removed to decreased the number of false negatives. The algorithm which uses images as input data performs the best and gives the fewest false positives and negatives. Based on these results the spatial information provided by the images is relevant to the classification algorithm.



(a) Visualization of the defective and functional cycles after K-means. (b) Visualization of the defective and functional devices after K-means.

Figure 6.19: Kmeans clustering of PCA using images as input data.



Performance of Kmeans based on device characterization

Figure 6.20: Performance of K-Means for different input data

6.3. Supervised learning

The algorithm is used to automatically classify defective and functional cycles, which are then linked to the corresponding device. We only have 49 devices which is too small of a dataset to split in a training and test set for such a complex problem. As a result, we will only use the supervised learning algorithms to classify the cycles of RRAM devices. These will then be linked back to the RRAM devices.

6.3.1. K-Nearest Neighbours

The performance of each input is shown in Figure 6.24. The results of the K-NN classifier in which the PCA analysis of the images are used as input are shown in Figure 6.21. Figure 6.22 shows the results in which the PCA analysis of the metrics is used as input and Figure 6.23 does this for the PCA analysis of the raw data.

Figure 6.21 shows that the cycles can be used as an indication to identify which devices are defective and which are not. Based on the results we set the tolerance of the number of defective cycles allowed per device at 30. If we link this to the results from the manual labelling then 87.8% of the devices can correctly be classified.



Figure 6.21: Number of defective cycles for each device using K-NN on image data.

For Figure 6.22 we set the tolerance at 80. This value is chosen based on the visual inspection of Figure 6.22. If we link this to the manual labels 79.6% of the devices are correctly classified.



Figure 6.22: Number of defective cycles for each device using K-NN on the metrics.

For the raw data as input set the accepted tolerance is set Figure 6.23 at 50 defective cycles for each device. Above that the devices are labelled as defective and below that the devices are labelled as functional. 81.6% of the devices are classified correctly using the K-NN algorithm on the raw data as input.



Figure 6.23: Number of defective cycles for each device using K-NN on raw data.

Figure 6.24 shows the performance of the K-NN algorithm using the different input sets. The images score the highest as 94% of the cycles are correctly labelled which translates to 84% of the devices. The raw data and metrics do not differ that much when looking at the number of devices that are correctly classified with 81.6% and 79.6% respectively. However, the number of cycles differs by 4% with 84% compared to 80% when the metrics are used as input.



Figure 6.24: Number of defective cycles for each device using K-NN on the metrics.

Figure 6.22, Figure 6.23 and Figure 6.21 show that all devices on word line zero with the exception of those on bitline zero and six can be considered functional devices.

6.3.2. Convolutional neural network

For the CNN we implemented the algorithm as described by [83]. The architecture is described in subsection 5.3.4. 80% of the cycles were used to train the convolutional neural model. 10% was used to test the CNN and 10% was used for validation. In total 20 epochs were used to train the convolutional neural neural network. After training for 20 epochs the CNN was validated using the validation dataset. The CNN classified 66.5% of the cycles correctly. To improve the functionality of the CNN we need to

improve the complexity of the CNN by adding additional layers, training for more epochs or using a larger sample size.

6.4. Summary

A statistical analysis was done on the HRS, LRS, Vset, Vreset and the Rratio. Furthermore, the dynamic time warping and euclidean algorithm is applied to a model I-V curve based on the manually labelled data. Three machine learning algorithms have been implemented: Kmeans, K-NN and a convolutional neural network. The statistical analysis of the individual metrics did not result in any viable results and none of the metrics were able to identify devices without detecting multiple false positives. V_{set} and V_{reset} are crucial metrics as the euclidean distance algorithm outperformed the dynamic time algorithm in which 81% of the devices are correctly classified compared to 70% for the DTW. The euclidean distance algorithm is not reliable enough to accurately classify RRAM devices but can be used as an initial data analysis for further processing. However, when the data is not labelled it is better to use the K-Means algorithm compared to the Model I-V curve and statistical analysis. The K-Means algorithm also performs best when the images are used as input data in which 90% of the cycles translating to 79.6% of the devices when the cycles are linked back to the device. The raw data and metrics perform worse with 71.4% and 59.6% respectively. First adding the principal components of each cycle for each device and then applying the Kmeans algorithm works as well as applying the algorithm on the individual cycles. 81.6% of the devices are correctly classified when the images are used as input. 71.6% when the raw data is used as input and 67.3% when the metrics are used as input. First combining the principal components for each device is better than linking the cycles back to the device afterwards. The K-NN algorithm classifies 94% of the cycles correctly and 84% of the devices when the images are used as input. The raw data and metrics do not differ that much when looking at the number of devices that are correctly classified with 81.6% and 79.6% respectively. The CNN classified 66.5% of the devices correctly. The image data as input scores the best and should be used when assessing the guality of the RRAM devices. If you have labelled data the K-NN is the best solution and should be used as the method is fast. However, the labelling of data is time consuming and can be challenging due to the device variability. In that case the K-means algorithm can be used for a quick analysis of the RRAM devices as it will already classify 90% of the cycles and 84% of the devices correctly. The CNN does not perform as well as expected but we think that if the CNN is optimized and tuned it can perform as well as the K-NN.

Discussion

This chapter discusses the validity of the work of this thesis work. First, the methods to extract the electrical parameters and the manual labelling of the device are discussed. After which the approach and the algorithms for RRAM classification are evaluated. The last section consists of the contribution of this masters work and suggestions for future research.

7.1. RRAM Data analytics

The RRAM data analysis as well as the statistical analysis of the results required the extraction of the electrical parameters: V_{set} , V_{reset} , HRS, LRS, $t_{H \Rightarrow L}$ and $t_{L \Rightarrow H}$. For all metrics the extraction technique was robust with the exception of a few edge cases. Additional, boundary conditions should be added to catch those boundary conditions as they affect the outlier detection algorithm as well as the K-means clustering. The method to extract the switching time is commonly done using an oscilloscope by applying pulsed voltage stresses. The electrical noise makes it challenging to determine a universal end point for the switching time for both from high resistive state to low resistive state and vice versa.

Furthermore, the data were not labelled. We designed a framework to label the individual cycles based on the current knowledge about the switching characteristics of RRAM devices. Kocak, Mitard, and Naskali did a similar approach in which the test data-set of MOSFETS was classified by researchers [83]. They concluded that the accuracy lies around 85% for a junior researcher and 92% for a expert researcher. To evaluate the performance of the classification algorithms we have also labelled the devices manually. Classifying RRAM devices using the opinion of a researcher is just as or maybe even more difficult than assessing the functionality of MOSFETs as we have less experience with RRAM. As a result, the method we have proposed to manually label RRAM cycles is not 100% accurate. Especially, because it is not yet fully understood how precisely a defect affect the electrical parameters. The manual labelling of the device, but additional data or devices with known defects should be evaluated using these methods to determine the performance of the algorithm.

7.2. Classification algorithm

The different classification algorithms provides an initial analysis of which devices are defective and which are not. The current method using the classification of individual I-V cycles which are linked to the devices. The cycles are classified independently from one another and the algorithm does not know to which device the cycle belongs. Ideally, the information about this is known and presented as a time series so that the information from other cycles of the same device are incorporated more efficiently in the classification of the RRAM device. However, the dataset consists of 49 RRAM devices are all from the same wafer. More devices from different wafers are required to validate the functionality of the classification algorithms.

7.3. Value of this work

This work presents a cheap method to automatically classify RRAM devices as defective or functional. The machine learning algorithms can be used as an initial classification algorithm to identify defective RRAM devices. After, this further characterization of the defective RRAM devices is required to enable the origin of the defect. For this more expensive methods such as optical characterization can be used or the visual examination of the individual I-V cycles. This work can be used as a foundation for further analysis of RRAM devices so that accurate defect models can be built for which new tests can be generated.

Additionally, the software used for this thesis work can be used to quickly characterize and identify defective RRAM devices. Further improvements of the classification methods or larger datasets to train the K-NN algorithm can improve the classification of RRAM devices. The classification method can be used to speed up the labelling of RRAM which can be used for the diagnosis of RRAM devices.

7.4. Future work

In the future, the work can be extended so that it can also identify which defect is present in the RRAM device. Currently, the algorithms only consists of two classes: defective and functional. Adding multiple classes for specific defects could make this possible. Additionally, the architecture of the CNN algorithm is directly copied from Kocak, Mitard, and Naskali and the training data we have is limited [83]. A more elaborate CNN architecture might be required consisting of additional layers, different tuning parameters or more epochs might give better results. However, due to time and the complexity of the CNN architecture this has not yet been explored completely and further analysis is required.

The electrical characterization of RRAM devices using machine learning looks promising and is much cheaper compared to optical characterization and memory tests. If in the future the defective devices can be linked to the underlying defect, this will lead to cheaper diagnosis of RRAM devices and allow the creation of accurate fault models.



Conclusion

This work focused on the automatic characterization of RRAM devices using electrical characterization. The software package allows the automatic identification of defective RRAM devices which is faster and cheaper than optical characterization. In the future the package could be elaborated so that it does not only indicate which device is defective but can also identify the defect that caused the abnormal behaviour.

Initially, the background information as well as a comparison is provided between the current and emerging memory technologies. Additionally, the manufacturing process of RRAM, the defects that can occur and how these defects affect the electrical parameters of RRAM are discussed. Then the stateof-the-art in defect identification and diagnosis is discussed in which a distinction was made between optical characterization, electrical characterization and memory testing.

The state-of-the-art highlighted the need for a cheap and cost-effective method to identify defects in RRAM devices. Optical characterization is expensive and in some cases destructive and memory tests are an abstraction of defects and are difficult to link back to the actual defect. Where electrical characterization has mainly been used to describe the behaviour of the RRAM device and how design choices affect this behaviour.

The electrical characterization has so far been done using the electrical parameters such as the switching voltage, switching time and the resistance in the high and low resistance state. Chapter 4 showed that the hardware specification were insufficient to identify defective RRAM devices and proposed a method to manually label RRAM cycles and devices which can be used for comparison and as a training set for supervised learning algorithms. The results from the manual labelling showed that 29 from the 49 devices did not behave as expected. The characterization of the electrical parameters in combination with the manual labelling of the RRAM cycles showed that there was a difference between the values of the electrical parameters for defective and functional RRAM devices.

Chapter 6 continued on this by doing a statistical analysis using the electrical parameters commonly used for RRAM characterization to identify defective RRAM devices. The analysis showed that the metrics were insufficient to identify defective devices and cannot account for the cycle-to-cycle variance of the RRAM device. The comparison between a model I-V curve and the test data using the euclidean distance resulted in 81% of the devices that can be correctly classified.

As the manual labelling is time consuming we also investigated the performance of the K-Means algorithm. When the I-V cycles were stored as images we showed that the algorithm can correctly classify 90% of the cycles correct and 79.6% of the devices are correctly classified. Furthermore, because we have labelled data we also evaluated the performance of two supervised algorithms: K-NN and a CNN. The K-NN is able to classify 90% of the cycles correctly which translates to 84% of the devices. The CNN classified 67% of the cycles correctly. The CNN has not yet been linked to the devices because the CNN has not been optimized yet and needs further improvement by changing the architecture of using more epochs before we can assess the performance of the CNN for the identification of defective RRAM devices.

This work forms the basis of using electrical tests to automatically identify defective RRAM devices and shows the relevance of machine learning algorithms in defect identification of RRAM.
Bibliography

- Seok Hee Lee. "Technology scaling challenges and opportunities of memory devices". In: *Technical Digest International Electron Devices Meeting, IEDM* (Jan. 2017), pp. 1.1.1–1.1.8. ISSN: 01631918. DOI: 10.1109/IEDM.2016.7838026.
- [2] Furqan Zahoor, Tun Zainal Azni Zulkifli, and Farooq Ahmad Khanday. "Resistive Random Access Memory (RRAM): an Overview of Materials, Switching Mechanism, Performance, Multilevel Cell (mlc) Storage, Modeling, and Applications". In: *Nanoscale Research Letters 2020 15:1* 15 (1 Apr. 2020), pp. 1–26. ISSN: 1556-276X. DOI: 10.1186/S11671-020-03299-9. URL: https: //link-springer-com.tudelft.idm.oclc.org/article/10.1186/S11671-020-03299-9.
- [3] Sung Kye Park. "Technology Scaling Challenge and Future Prospects of DRAM and NAND Flash Memory". In: 2015 IEEE 7th International Memory Workshop, IMW 2015 (July 2015). DOI: 10. 1109/IMW.2015.7150307.
- [4] Shigeru Shiratake. "Scaling and Performance Challenges of Future DRAM". In: 2020 IEEE International Memory Workshop, IMW 2020 - Proceedings (May 2020). DOI: 10.1109/IMW48823. 2020.9108122.
- [5] Ashwani Kumar Yadav et al. "Various Issues and considerations for the Static Power Consumption in NANO-CMOS: Design Perspective". In: *Materials Today: Proceedings* 10 (Jan. 2019), pp. 136–141. ISSN: 2214-7853. DOI: 10.1016/J.MATPR.2019.02.198.
- [6] Jagan Singh Meena et al. "Overview of emerging nonvolatile memory technologies". In: Nanoscale Research Letters 9 (1 Sept. 2014), pp. 1–33. ISSN: 1556276X. DOI: 10.1186/1556-276X-9-526/FIGURES/29. URL: https://link-springer-com.tudelft.idm.oclc.org/ article/10.1186/1556-276X-9-526.
- Shimeng Yu and Pai Yu Chen. "Emerging Memory Technologies: Recent Trends and Prospects". In: *IEEE Solid-State Circuits Magazine* 8 (2 Mar. 2016), pp. 43–56. ISSN: 19430582. DOI: 10. 1109/MSSC.2016.2546199.
- [8] Janice H. Nickel et al. "Memristor structures for high scalability: Non-linear and symmetric devices utilizing fabrication friendly materials and processes". In: *Microelectronic Engineering* 103 (Mar. 2013), pp. 66–69. ISSN: 0167-9317. DOI: 10.1016/J.MEE.2012.09.007.
- [9] L. M.Bolzani Poehls et al. "Review of Manufacturing Process Defects and Their Effects on Memristive Devices". In: Journal of Electronic Testing: Theory and Applications (JETTA) 37 (4 Aug. 2021), pp. 427–437. ISSN: 15730727. DOI: 10.1007/S10836-021-05968-8/FIGURES/4. URL: https://link-springer-com.tudelft.idm.oclc.org/article/10.1007/ s10836-021-05968-8.
- Shimeng Yu et al. "Compute-in-Memory Chips for Deep Learning: Recent Trends and Prospects".
 In: *IEEE Circuits and Systems Magazine* 21 (3 July 2021), pp. 31–56. ISSN: 15580830. DOI: 10.1109/MCAS.2021.3092533.
- [11] Amin Shafiee, Sudeep Pasricha, and Mahdi Nikdast. "A Survey on Optical Phase-Change Memory: The Promise and Challenges". In: *IEEE Access* 11 (2023), pp. 11781–11803. ISSN: 21693536. DOI: 10.1109/ACCESS.2023.3241146.
- [12] Pinaki Mazumder, Sung Mo Kang, and Rainer Waser. "Memristors: Devices, models, and applications". In: *Proceedings of the IEEE* 100 (6 2012), pp. 1911–1919. ISSN: 00189219. DOI: 10.1109/JPROC.2012.2190812.
- [13] Michael L. Bushnell and Vishwani D. Agrawal. "Essentials of Electronic Testing for Digital, Memory and Mixed-Signal VLSI Circuits". In: 17 (2002). DOI: 10.1007/B117406. URL: http: //link.springer.com/10.1007/b117406.

- [14] Moritz Fieback, Mottaqiallah Taouil, and Said Hamdioui. "Testing Resistive Memories: Where are We and What is Missing?" In: *Proceedings - International Test Conference* 2018-October (Jan. 2019). ISSN: 10893539. DOI: 10.1109/TEST.2018.8624895.
- [15] Moritz Fieback et al. "Defects, Fault Modeling, and Test Development Framework for RRAMs". In: ACM Journal on Emerging Technologies in Computing Systems 18 (3 Apr. 2022), p. 52. ISSN: 15504840. DOI: 10.1145/3510851. URL: https://doi.org/10.1145/3510851.
- [16] Elena Ioana Vatajelu et al. "Challenges and solutions in emerging memory testing". In: IEEE Transactions on Emerging Topics in Computing 7 (3 July 2019), pp. 493–506. ISSN: 21686750. DOI: 10.1109/TETC.2017.2691263.
- [17] Nor Zaidi Haron and Said Hamdioui. "On defect oriented testing for hybrid CMOS/memristor memory". In: *Proceedings of the Asian Test Symposium* (2011), pp. 353–358. ISSN: 10817735. DOI: 10.1109/ATS.2011.66.
- [18] L. M. Bolzani Poehls. "Embedded Tutorial RRAMs: How to Guarantee Their Quality Test after Manufacturing?" In: *Proceedings - 2023 26th International Symposium on Design and Diagnostics of Electronic Circuits and Systems, DDECS 2023* (2023), pp. 167–168. DOI: 10.1109/ DDECS57882.2023.10139525.
- [19] Mario Lanza et al. "Recommended Methods to Study Resistive Switching Devices". In: Advanced Electronic Materials 5 (1 Jan. 2019). ISSN: 2199160X. DOI: 10.1002/AELM.201800143.
- [20] A. Fantini et al. "Lateral and vertical scaling impact on statistical performances and reliability of 10nm TiN/Hf(AI)O/Hf/TiN RRAM devices". In: Digest of Technical Papers - Symposium on VLSI Technology (Sept. 2014). ISSN: 07431562. DOI: 10.1109/VLSIT.2014.6894433.
- [21] Alessandro Grossi et al. "Performance and reliability comparison of 1T-1R RRAM arrays with amorphous and polycrystalline HfO2". In: 2016 Joint International EUROSOI Workshop and International Conference on Ultimate Integration on Silicon, EUROSOI-ULIS 2016 (Mar. 2016), pp. 80–83. DOI: 10.1109/ULIS.2016.7440057.
- [22] Ching Yi Chen et al. "RRAM defect modeling and failure analysis based on march test and a novel squeeze-search scheme". In: *IEEE Transactions on Computers* 64 (1 Jan. 2015), pp. 180–190. ISSN: 00189340. DOI: 10.1109/TC.2014.12.
- [23] Peng Liu et al. "Efficient March test algorithm for 1T1R cross-bar with complete fault coverage". In: Electronics Letters 52 (18 Sept. 2016), pp. 1520–1522. ISSN: 1350-911X. DOI: 10.1049/ EL.2016.1693. URL: https://onlinelibrary.wiley.com/doi/full/10.1049/el. 2016.1693%20https://onlinelibrary.wiley.com/doi/abs/10.1049/el.2016. 1693%20https://ietresearch.onlinelibrary.wiley.com/doi/10.1049/el. 2016.1693.
- [24] Sachhidh Kannan, Ramesh Karri, and Ozgur Sinanoglu. "Sneak path testing and fault modeling for multilevel memristor-based memories". In: 2013 IEEE 31st International Conference on Computer Design, ICCD 2013 (2013), pp. 215–220. DOI: 10.1109/ICCD.2013.6657045.
- [25] Moritz Fieback et al. "Device-aware test: A new test approach towards DPPB level". In: Proceedings International Test Conference 2019-November (Nov. 2019). ISSN: 10893539. DOI: 10.1109/ITC44170.2019.9000134.
- [26] Jonghan Kwon et al. "In situ biasing TEM investigation of resistive switching events in TiO 2based RRAM". In: IEEE International Reliability Physics Symposium Proceedings (2014). ISSN: 15417026. DOI: 10.1109/IRPS.2014.6860680.
- [27] J. Sun et al. "Real time observation of nanoscale multiple conductive filaments in RRAM by using advanced in-situ TEM". In: Proceedings of the International Symposium on the Physical and Failure Analysis of Integrated Circuits, IPFA (2013), pp. 560–562. DOI: 10.1109/IPFA.2013. 6599223.
- [28] Chaolun Wang and Xing Wu. "Analysis of nano-filament evolution in Ni-based RRAM devices using in-situ TEM". In: 2016 13th IEEE International Conference on Solid-State and Integrated Circuit Technology, ICSICT 2016 - Proceedings (July 2017), pp. 876–879. DOI: 10.1109/ICSICT. 2016.7999067.

- [29] K. L. Pey et al. "Understanding the switching mechanism in RRAM using in-situ TEM". In: 2016 IEEE Silicon Nanoelectronics Workshop, SNW 2016 (Sept. 2016), pp. 36–37. DOI: 10.1109/ SNW.2016.7577973.
- [30] Sanjoy Kumar Nandi et al. "Effect of electrode roughness on electroforming in HfO2 and defectinduced moderation of electric-field enhancement". In: *Physical Review Applied* 4 (6 Dec. 2015), p. 064010. ISSN: 23317019. DOI: 10.1103 / PHYSREVAPPLIED.4.064010 / FIGURES / 13 / MEDIUM. URL: https://journals.aps.org/prapplied/abstract/10.1103 / PhysRevApplied.4.064010.
- [31] C. Charpin-Nicolle et al. "Impact of roughness of TiN bottom electrode on the forming voltage of HfO2 based resistive memories". In: *Microelectronic Engineering* 221 (Jan. 2020), p. 111194. ISSN: 0167-9317. DOI: 10.1016/J.MEE.2019.111194.
- [32] M. Lanza et al. "Grain boundaries as preferential sites for resistive switching in the HfO 2 resistive random access memory structures". In: *Applied Physics Letters* 100 (12 Mar. 2012), p. 123508. ISSN: 00036951. DOI: 10.1063/1.3697648/23899. URL: /aip/apl/article/100/12/ 123508/23899/Grain-boundaries-as-preferential-sites-for.
- [33] Danijela Efnusheva, Ana Cholakoska, and Aristotel Tentov. "A Survey of Different Approaches for Overcoming the Processor - Memory Bottleneck". In: *International Journal of Computer Science and Information Technology* 9 (2 Apr. 2017). Figure 1: Memory vs performance gap
br/>, pp. 151–163. ISSN: 09754660. DOI: 10.5121/IJCSIT.2017.9214.
- [34] R. Iyer et al. "Advances in Microprocessor Cache Architectures Over the Last 25 Years". In: IEEE Micro 41.06 (Nov. 2021), pp. 78–88. ISSN: 1937-4143. DOI: 10.1109/MM.2021.3114903.
- [35] Carlos Carvalho. "The Gap between Processor and Memory Speeds". In: ().
- [36] David A Patterson and John L Hennessy. Computer Organization and Design: The Hardware/software Interface (The Morgan Kaufmann Series in Computer Architecture and Design). 2014, p. 800. ISBN: 9780123747501. URL: https://books.google.com/books?id=G7IMAwAAQBAJ&dq= Computer+Organization+and+Design:+The+Hardware/software+Interface+ fifth+edition&hl=en&sa=X&ved=OahUKEwjM7o-F5K7XAhWMAMAKHR1iBK0Q6AEILTAB.
- [37] Suzanne J. Matthews, Tia Newhall, and Kevin C. Webb. "Dive into systems : a gentle introduction to computer systems". In: ().
- [38] Cache and Memory Hierarchy Design: A Performance Directed Approach Steven A. Przybylski - Google Books. URL: https://books.google.nl/books?hl=en&lr=&id=G-D6KFwnVsgC&oi=fnd&pg=PP1&dq=memory+hierarchy+book&ots=JSZ42HZqD8&sig= XOZc_mfPY_oE6m9EegC7aLFOh_E&redir_esc=y#v=onepage&q=memory%20hierarchy% 20book&f=false.
- [39] Bruce Jacob, Spencer W. Ng, and David T. Wang. "Memory Systems: Cache, DRAM, Disk". In: Memory Systems: Cache, DRAM, Disk (Jan. 2007), pp. 1–982. DOI: 10.1016/B978-0-12-379751-3.X5001-2.
- [40] Roberto Bez et al. "Introduction to flash memory". In: Proceedings of the IEEE 91 (4 2003), pp. 489–501. ISSN: 00189219. DOI: 10.1109/JPROC.2003.811702. URL: https://www. researchgate.net/publication/2986127_Introduction_to_Flash_memory.
- [41] Thomas M. Coughlin. "Digital Storage in Consumer Electronics". In: *Digital Storage in Consumer Electronics* (2018). DOI: 10.1007/978-3-319-69907-3.
- [42] Ahmed Eltawil et al. "AS8-static random access memory (SRAM): Asymmetric SRAM architecture for soft error hardening enhancement". In: *IET Circuits, Devices and Systems* 11 (Apr. 2016). DOI: 10.1049/iet-cds.2015.0318.
- [43] Wenchao Liu et al. "A Trustworthy Key Generation Prototype Based on DDR3 PUF for Wireless Sensor Networks". In: Sensors (Basel, Switzerland) 14 (July 2014), pp. 11542–11556. DOI: 10. 3390/s140711542.
- [44] T Windbacher. "Engineering gate stacks for Field-Effect-Transistors". In: (Dec. 2022).

- [45] Amol S. Sankpal and D. J. Pete. "Study and Analysis of Leakage current and leakage power in 1T1C DRAM at Nano Scale Technology". In: 2020 4th International Conference on Electronics, Communication and Aerospace Technology (ICECA). 2020, pp. 99–104. DOI: 10.1109/ ICECA49313.2020.9297568.
- [46] Jonathan Yu and Koorosh Aflatooni. "Leakage Current in DRAM Memory Cell". In: 2006 16th Biennial University/Government/Industry Microelectronics Symposium. 2006, pp. 191–194. DOI: 10.1109/UGIM.2006.4286380.
- [47] R. Zhao et al. "Study of Phase Change Random Access Memory (PCRAM) at the nano-scale". In: *Proceedings - 2007 Non-Volatile Memory Technology Symposium, NVMTS 07* (2007), pp. 36– 39. DOI: 10.1109/NVMT.2007.4389941.
- [48] H. S.Philip Wong et al. "Phase change memory". In: *Proceedings of the IEEE* 98 (12 2010), pp. 2201–2227. ISSN: 00189219. DOI: 10.1109/JPROC.2010.2070050.
- [49] A. V. Khvalkovskiy et al. "Erratum: Basic principles of STT-MRAM cell operation in memory arrays". In: Journal of Physics D: Applied Physics 46 (13 Feb. 2013), p. 139601. ISSN: 0022-3727. DOI: 10.1088/0022-3727/46/13/139601. URL: https://iopscience.iop.org/article/10.1088/0022-3727/46/13/139601%20https://iopscience.iop.org/article/10.1088/0022-3727/46/13/139601/meta.
- [50] Tetsuo Endoh and Hiroaki Honjo. "A recent progress of spintronics devices for integrated circuit applications". In: *Journal of Low Power Electronics and Applications* 8 (4 2018). ISSN: 20799268. DOI: 10.3390/JLPEA8040044.
- [51] H. S.Philip Wong et al. "Metal-oxide RRAM". In: *Proceedings of the IEEE* 100 (6 2012), pp. 1951– 1970. ISSN: 00189219. DOI: 10.1109/JPROC.2012.2190369.
- [52] Tijs Hol. Modeling the physics of RRAM defects: A model simulating RRAM defects on a macroscopic physical level. 2020. URL: https://repository.tudelft.nl/islandora/ object/uuid%3A968aa2e8-042e-437b-a7b7-d26835067757.
- [53] V. Iglesias et al. "Dielectric breakdown in polycrystalline hafnium oxide gate dielectrics investigated by conductive atomic force microscopy". In: *Journal of Vacuum Science and Technology B* 29.1 (Jan. 2011), 01AB02. ISSN: 2166-2746. DOI: 10.1116/1.3532945. eprint: https: //pubs.aip.org/avs/jvb/article-pdf/doi/10.1116/1.3532945/15882094/ 01ab02_1_online.pdf. URL: https://doi.org/10.1116/1.3532945.
- [54] Dieter Wolf. "Structure and Energy of Grain Boundaries". In: Handbook of Materials Modeling: Methods. Ed. by Sidney Yip. Dordrecht: Springer Netherlands, 2005, pp. 1953–1983. ISBN: 978-1-4020-3286-8. DOI: 10.1007/978-1-4020-3286-8_102. URL: https://doi.org/10. 1007/978-1-4020-3286-8_102.
- [55] C. Nguyen et al. "Study of forming impact on 4Kbit RRAM array performances and reliability". In: 2017 IEEE 9th International Memory Workshop, IMW 2017 (June 2017). DOI: 10.1109/IMW. 2017.7939105.
- [56] Jan M. Rabaey. Digital Integrated Circuits: A Design Perspective. USA: Prentice-Hall, Inc., 1996. ISBN: 0131786091.
- [57] Kelin J. Kuhn et al. "Process technology variation". In: *IEEE Transactions on Electron Devices* 58 (8 Aug. 2011), pp. 2197–2208. ISSN: 00189383. DOI: 10.1109/TED.2011.2121913.
- [58] D. lelmini and R. Waser. Resistive Switching: From Fundamentals of Nanoionic Redox Processes to Memristive Device Applications. Wiley, 2015. ISBN: 9783527680931. URL: https://books. google.nl/books?id=E21cCwAAQBAJ.
- [59] Said Hamdioui. "Testing Static Random Access Memories". In: 26 (2004). DOI: 10.1007/978-1-4757-6706-3. URL: http://link.springer.com/10.1007/978-1-4757-6706-3.
- [60] Ad J. van de Goor and Zaid Al-Ars. "Functional memory faults: a formal notation and a taxonomy". In: Proceedings of the IEEE VLSI Test Symposium (2000), pp. 281–289. DOI: 10.1109/VTEST. 2000.843856.
- [61] Yong-Xiao Chen and Jin-Fu Li. "Fault modeling and testing of 1T1R memristor memories". In: 2015 IEEE 33rd VLSI Test Symposium (VTS). 2015, pp. 1–6. DOI: 10.1109/VTS.2015. 7116247.

- [62] Seyed Nima Mozaffari, Spyros Tragoudas, and Themistoklis Haniotakis. "More Efficient Testing of Metal-Oxide Memristor-Based Memory". In: *IEEE Transactions on Computer-Aided Design* of Integrated Circuits and Systems 36.6 (2017), pp. 1018–1029. DOI: 10.1109/TCAD.2016. 2608863.
- [63] O. Ginez, J.-M. Portal, and Ch. Muller. "Design and Test Challenges in Resistive Switching RAM (ReRAM): An Electrical Model for Defect Injections". In: 2009 14th IEEE European Test Symposium. 2009, pp. 61–66. DOI: 10.1109/ETS.2009.23.
- [64] N.Z. Haron and Said Hamdioui. "DfT schemes for resistive open defects in RRAMs". In: Mar. 2012, pp. 799–804. ISBN: 978-1-4577-2145-8. DOI: 10.1109/DATE.2012.6176603.
- [65] Sachhidh Kannan et al. "Sneak-path testing of memristor-based memories". English (US). In: Proceedings - 26th International Conference on VLSI Design, VLSID 2013 - Concurrently with 12th International Conference on Embedded Systems Design, ES 2013. Proceedings of the IEEE International Conference on VLSI Design. 2013 26th International Conference on VLSI Design, VLSID 2013 and 12th International Conference on Embedded Systems, ES 2013 ; Conference date: 05-01-2013 Through 10-01-2013. 2013, pp. 386–391. ISBN: 9780769548890. DOI: 10. 1109/VLSID.2013.219.
- [66] Sachhidh Kannan et al. "Sneak-path testing of crossbar-based nonvolatile random access memories". English (US). In: *IEEE Transactions on Nanotechnology* 12.3 (2013), pp. 413–426. ISSN: 1536-125X. DOI: 10.1109/TNANO.2013.2253329.
- [67] Said Hamdioui and Ad J. van de Goor. "Experimental analysis of spot defects in SRAMs: Realistic fault models and tests". In: *Proceedings of the Asian Test Symposium* (2000), pp. 131–138. ISSN: 10817735. DOI: 10.1109/ATS.2000.893615.
- [68] Kamran Zarrineh, Aneesha P. Deo, and R. Dean Adams. "Defect analysis and realistic fault model extensions for static random access memories". In: *Records of the IEEE International Workshop* on Memory Technology, Design and Testing (2000), pp. 119–124. ISSN: 10874852. DOI: 10. 1109/MTDT.2000.868625.
- [69] A. Fantini et al. "Intrinsic switching variability in HfO2 RRAM". In: 2013 5th IEEE International Memory Workshop, IMW 2013 (2013), pp. 30–33. DOI: 10.1109/IMW.2013.6582090.
- [70] Moritz Fieback et al. "Intermittent Undefined State Fault in RRAMs". In: Proceedings of the European Test Workshop 2021-May (May 2021). ISSN: 15581780. DOI: 10.1109/ETS50041. 2021.9465401.
- [71] Wenhong Wang et al. "Memristive behavior of ZnO/Au film investigated by a TiN CAFM Tip and its model based on the experiments". In: *IEEE Transactions on Nanotechnology* 11 (6 2012), pp. 1135–1139. ISSN: 1536125X. DOI: 10.1109/TNANO.2012.2214486.
- [72] M. Lanza et al. "High-Performance Piezoelectric Nanogenerators Using Two-Dimensional Flexible Top Electrodes". In: Advanced Materials Interfaces 1 (5 Aug. 2014), p. 1300101. ISSN: 2196-7350. DOI: 10.1002/ADMI.201300101. URL: https://onlinelibrary-wileycom.tudelft.idm.oclc.org/doi/full/10.1002/admi.201300101%20https: //onlinelibrary-wiley-com.tudelft.idm.oclc.org/doi/abs/10.1002/admi. 201300101%20https://onlinelibrary-wiley-com.tudelft.idm.oclc.org/doi/ 10.1002/admi.201300101.
- [73] Chikako Yoshida et al. "Direct observation of oxygen movement during resistance switching in NiO/Pt film". In: Applied Physics Letters 93 (4 July 2008), p. 42106. ISSN: 00036951. DOI: 10. 1063/1.2966141/324177. URL: /aip/apl/article/93/4/042106/324177/Directobservation-of-oxygen-movement-during.
- [74] S. Chouhan et al. "Effect of thermal resistance and scaling on dc-IV characteristics of PCMO based RRAM devices". In: *Device Research Conference - Conference Digest, DRC* (Aug. 2017). ISSN: 15483770. DOI: 10.1109/DRC.2017.7999428.
- [75] D. Maldonado et al. "Variability estimation in resistive switching devices, a numerical and kinetic Monte Carlo perspective". In: *Microelectronic Engineering* 257 (Mar. 2022), p. 111736. ISSN: 0167-9317. DOI: 10.1016/J.MEE.2022.111736.

- [76] D. Maldonado et al. "Parameter extraction techniques for the analysis and modeling of resistive memories". In: *Microelectronic Engineering* 265 (Sept. 2022), p. 111876. ISSN: 0167-9317. DOI: 10.1016/J.MEE.2022.111876.
- [77] D. Maldonado et al. "Experimental study of the series resistance effect and its impact on the compact modeling of the conduction characteristics of HfO2-based resistive switching memories". In: Journal of Applied Physics 130 (5 Aug. 2021), p. 54503. ISSN: 10897550. DOI: 10.1063/5. 0055982/158776. URL: /aip/jap/article/130/5/054503/158776/Experimentalstudy-of-the-series-resistance-effect.
- [78] C. Acal et al. "Phase-type distributions for studying variability in resistive memories". In: Journal of Computational and Applied Mathematics 345 (Jan. 2019), pp. 23–32. ISSN: 0377-0427. DOI: 10.1016/J.CAM.2018.06.010.
- [79] Shibing Long et al. "Analysis and modeling of resistive switching statistics". In: Journal of Applied Physics 111 (7 Apr. 2012), p. 74508. ISSN: 00218979. DOI: 10.1063/1.3699369/391166. URL: /aip/jap/article/111/7/074508/391166/Analysis-and-modeling-ofresistive-switching.
- [80] H. Y. Lee et al. "Evidence and solution of over-RESET problem for HfOX based resistive memory with sub-ns switching speed and high endurance". In: 2010 International Electron Devices Meeting. 2010, pp. 19.7.1–19.7.4. DOI: 10.1109/IEDM.2010.5703395.
- [81] David Clark-Carter. "Interquartile Range". In: Oct. 2005. ISBN: 0470860804. DOI: 10.1002/ 0470013192.bsa311.
- [82] Daniel Martin et al. "ScanGAN360: A Generative Model of Realistic Scanpaths for 360° Images". In: (Mar. 2021).
- [83] Husnu Murat Kocak, Jerome Mitard, and Ahmet Teoman Naskali. "Combined Machine Learning Techniques for Characteristics Classification and Threshold Voltage Extraction of Transistors".
 In: IEEE International Conference on Microelectronic Test Structures 2022-March (2022). DOI: 10.1109/ICMTS50340.2022.9898251.
- [84] Miloš Radovanović, Alexandros Nanopoulos, and Mirjana Ivanović. "Nearest Neighbors in High-Dimensional Data: The Emergence and Influence of Hubs". In: Proceedings of the 26th Annual International Conference on Machine Learning. ICML '09. Montreal, Quebec, Canada: Association for Computing Machinery, 2009, pp. 865–872. ISBN: 9781605585161. DOI: 10.1145/ 1553374.1553485. URL: https://doi-org.tudelft.idm.oclc.org/10.1145/ 1553374.1553485.
- [85] Michael Greenacre et al. "Principal component analysis". In: 2 (Dec. 2022), p. 100. DOI: 10. 1038/s43586-022-00184-w.
- [86] Sidharth Mishra et al. "Principal Component Analysis". In: International Journal of Livestock Research (Jan. 2017), p. 1. DOI: 10.5455/ijlr.20170415115235.
- [87] Sasan Karamizadeh et al. "An Overview of Principal Component Analysis". In: *Journal of Signal and Information Processing* (Aug. 2013). DOI: 10.4236/jsip.2013.43B031.
- [88] F. Pedregosa et al. "Scikit-learn: Machine Learning in Python". In: Journal of Machine Learning Research 12 (2011), pp. 2825–2830.